

Surroundings: history-dependent optimization and applications to circular and sustainable economics

Lisa Morhaim, Ayşegül Yıldız Ulus

▶ To cite this version:

Lisa Morhaim, Ayşegül Yıldız Ulus. Surroundings: history-dependent optimization and applications to circular and sustainable economics: CRED WORKING PAPER 2024-08. 2024. hal-04606317

HAL Id: hal-04606317 https://univ-pantheon-assas.hal.science/hal-04606317

Preprint submitted on 10 Jun 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License



CRED Centre de recherche en économie et droit

CRED WORKING PAPER n^o 2024-08

Surroundings: history-dependent optimization and applications to circular and sustainable economics

June, 2024

LISA MORHAIM^{*} AYSEGÜL YILDIZ ULUS[†]

*Université Paris-Panthéon-Assas, CRED, F-75005, Paris, France.

[†]Galatasaray University, Department of Mathematics

SURROUNDINGS: HISTORY-DEPENDENT OPTIMIZATION AND APPLICATIONS TO CIRCULAR AND SUSTAINABLE ECONOMICS

Lisa Morhaim^{*} and Ayşegül Yıldız Ulus[†]

Friday 7th June, 2024

Abstract

Decisions and ideas arise within "surroundings", the set of natural and cultural conditions in which organisms live and human activities take place. Moreover, current decisions are dependent to the past. In this paper, we propose a framework which incorporates general history-dependencies and surroundings. It is designed to take into account both that history dependencies affect the surroundings and that history dependencies are affected by the surroundings. These are crucial issues in many domains, including Economics, Law, Artificial Intelligence and Reinforcement Learning. To be able to consider all kinds of phenomena and issues from different areas, history-dependencies are modeled in a very general way through the introduction of a memory function (not necessarily *consumption* history formation processes as previously in the economic literature). Our modeling is tractable, interpretable within many diverse contexts and allows several simultaneous history dependencies. We obtain optimization results and develop dynamic programming tools to deal with such models, in particular, we show that there exists a solution and the value function is the unique fixed point of the Bellman operator. Since environmental and sustainable variables are influenced by the memory of our past decisions and can be taken as surroundings, as a by-product, we furthermore introduce a very general sustainable framework which fits many existing environmental and sustainable models including circular economy models. It provides a basis for future environmental analysis.

KEYWORDS: History-dependent model, Surroundings, Circular Economy, Environment, Pollution, Sustainability, Growth, Habits, Satiation, Optimal management of natural resources, Law, Optimal growth, Intertemporal decisions with instantaneous history-dependencies, Dynamic programming, Reinforcement Learning, Artificial Intelligence.

JEL CLASSIFICATION: C61, D90, Q01, Q5. MSC CLASSIFICATION: 90C39, 68T05, 91B55.

^{*}Université Paris-Panthéon-Assas, CRED Paris Center for Law and Economics, lisa.morhaim@u-paris2.fr †Galatasaray University, Department of Mathematics, aulus@gsu.edu.tr.

This work was supported by the French Cooperation with Galatasaray University Consortium (*Thematic Project 2023-2024: Society and Culture: Decision and Transitions*) and also by the Galatasaray University Research Grant Number FBA-2024-1206. We are grateful to Rabah Amir for useful discussions and comments during this work. We thank Manjira Datta, Kevin Reffett and Juan-Pablo Rincon Zapatero for having outreached important questions about this work. We also thank Lucasz Wozny, Agnieszka Wiszniewska-Matyszkiel, Jean-Pierre Drugeon, Luciano de Castro, Dominika Machowska and the SAET 2023 participants. Also a special thank for interesting exchanges with Ayşe Sevencan; the members of CRED; Youcef Askoura, Antoine Billot, Sandra Cavaco, Chantal Marlats, Lucie Ménager and the LEMMA Seminar participants; Philippe Bich, Jean-Marc Bonnisseau, Pascal Gourdel, Bernard de Meyer and Agnieszka Rusinowska at CES; Jérémie Cabessa, Yann Strozecki and the participants to the DAVID-Université Versailles-Saint-Quentin Seminar.

1	INTRODUCTION	2
2	A GENERAL HISTORY-DEPENDENT FRAMEWORK (GHDF) 2.1 The general history-dependent model 2.1.1 Notations and feasible sets 2.1.2 Assumptions 2.1.3 The objective is well-defined 2.1.4 Existence and uniqueness of the solution 2.1.5 The value function and Bellman equation 2.1.6 Properties of the value function 2.1.7 The value function is the unique fixed-point of the Bellman operator 2.1.2 An easily implementable history-dependent model (EIHDF)	5 5 7 8 8 8 8 9 9
	 2.2.1 The problem	10 10 10 11 11
3	 APPLICATION: A GENERAL SUSTAINABLE FRAMEWORK (GSF) 3.1 The model	12 12 14 15
4	 SOME MODELS AND EXAMPLES 4.1 History-dependent optimal growth models with (consumption) habit formation or satiation 4.2 History-dependent optimization models with environmental effects 4.3 Optimal management of natural resources 4.4 Circular economy (CE) and circular and cumulative causation (CCC) models 4.4.1 CE models without recycling habits 4.4.2 CE models with recycling habits 4.4.3 CCC models 4.4.4 On the circular and sustainable economy research agenda 	 16 17 19 20 20 21 21 22
5	CONCLUSION	23
6	APPENDIX 6.1 Proof of Proposition 2.1	24 24 24 26 26 27

1 Introduction

Our decisions and ideas arise within surroundings. Surroundings are the set of natural and cultural conditions in which organisms live and human activities take place, and play a crucial role in many areas, whether in History, Philosophy, Epistemology, Sociology, Economy, Law or Artificial intelligence. Surroundings' evolutions and human activities' histories are mutually dependent. We aim to build frameworks in which the *decisions* and *surroundings* may interlink in various ways. This is done by modeling their history evolutions and memory formations. While Goldsmith and Laks study the evolution of *ideas* in their area and refer to the way they abstract or not from human and social context, "The first is sometimes called *internal history*, the second *external history*; both are important for us." (Goldsmith and Laks[21]), we approach such issues in economics, considering that both decisions and surroundings must be taken into account, simultaneously or not.

The literature increasingly reveals the importance of history dependencies in the decisions' processes. Mainly, the instantaneous decision criteria crucially depend on the sequences of past events (which include, but are not restricted to, past decisions). It still remains to understand the historydependencies formations, to highlight how history dependencies affect our decisions, and how they are affected by our decisions.

In Economics, history dependencies have mainly appeared in models with a representative consumer taking her own past consumptions as a reference point. Axiomatically (Rozen[44], Rustichini and Siconolfi[45, 46], He Dyer Butler[24], etc), theoretically (Ryder and Heal[25], Caroll, Overland and Weil[11], Bambi and Gozzi[6], Bambi, Ghilli, Gozzi and Leocata[5], etc) and empirically (Havranek, Rusnak and Sokolova[22], Fuhrer[18], Ravina[42], Dynan[15], etc), different viewpoints have contributed to explain some observed phenomena. In this literature, a current history level is introduced and several ways of history formation from consumptions have been considered. Morhaim and Ulus[40] proposed a unified framework allowing very general consumption history formation processes, opening a path to study various history-dependencies issues (such as habits or satiation, but also addiction and many other effects) while providing results and dynamic programming tools to deal with these models.

Yet some important issues remain to consider, among which the two following crucial ones. Firstly, history formation processes of the reference points arise, that are not necessarily *consumption* history formation processes. New modellings may be designed to incorporate different kinds of history-dependencies. Secondly, our decisions are affected by the surroundings in which they occur. By our definition of surroundings, they might be whether natural or cultural conditions, such as legal frameworks, (social or cultural) norms, environment, etc. They thus include but are not restricted to environmental and sustainable variables. It is important to understand, on one hand how do history dependencies affect the surroundings, and on the other hand how history dependencies are affected by the surroundings. These two important issues are addressed in this paper.

As a by-product, our framework in this paper fits many environmental and sustainable models, including circular economy models. Indeed, environmental and sustainable variables are influenced by (the memory of our past) decisions and can be taken as surroundings. This leads us to introduce a very general sustainable framework, which provides a basis for future environmental analysis.

Circular economy (CE) is an important issue on current policy agendas worldwide (Abad-Segura, de la Fuente, González-Zamar and Belmonte-Ureña[1], Fitch-Roy, Benson and Monciardini [17], de Melo, de Oliveira, de Sousa, Vieira and Amaral[12]), CE aiming to increase the efficiency of resources use to achieve a better balance and harmony between economy, environment and society (Ghisellini, Cialani and Ulgiati[20]). Among scholars, such discussions and ideas date at least back

to the second half of the the twentieth century (Boulding[9], Pearce and Turner [41]). The conceptual and theoretical understandings as well as CE strategies and implementations are still currently discussed (Andersen[3], Kirchherr, Reike and Hekkert[32], Kalmykova Sadagopan and Rosado[27]). Nevertheless, CE is most frequently depicted as a combination of reduce, reuse and recycle activities (Kirchherr, Reike and Hekkert[32]), but also design, implying a focus on the entire life cycle of the processes the interaction between the process and the environment and the economy in which it is embedded (Ghisellini, Cialani and Ulgiati[20]). Although not systematically, a large strand of research analyze CE interlinked with sustainable development (Ghisellini, Cialani and Ulgiati[20], Ritzén and Sandström[43]. The Ellen Mac Arthur Foundation Report[37] thus aimed to demonstrate how circular economy principles and strategies significantly reduce greenhouse gas emissions and calls for integrating efforts to respond to climate change with those to accelerate the transition to a circular economy.

We now detail the way we address history-dependencies, history formation and surroundings in this paper. We provide a framework with general history formations. This allows to consider the various kinds of history-dependencies that are currently taken into account in the economic literature. We introduce a *memory function* to model the history formation process. The framework in Morhaim and Ulus[40], in which *consumption* history is considered becomes a particular case of transforming the decisions into a history sequence. Indeed, this is a case in which *consumption* is kept in memory, which is a function of previous date and current capital stock decisions. It means keeping in memory a particular function of these decisions. Our general history-dependent framework allows, not only to keep in memory *the particular function* defining consumption, but *any function* of the previous date and current decisions. Thus, Morhaim and Ulus[40] model as well as all its examples¹ are particular cases of our general history-dependent framework.

Furthermore, the framework we propose is conceived to take into account various kinds and widely interpretable surroundings. They can model habits (Ryder and Heal[25], Rozen[44], Rustichini and Siconolfi[46], Caroll, Overland and Weil[10]), satiation (He, Dyer and Butler[23]), environmental issues and pollution (Ikefuji[26], Löfgren[36], Van Der Ploeg and Withagen[53]), optimal management of natural resource (Smulders, Toman and Withagen[48], Ulus[51]), sustainability issues and circular economy (CE) models (George, Chi-ang Lin and Chen[19], Kasioumi[28, 29, 31]), and circular and causation (CCC) models (Donaghy[13, 14]), but also many other phenomena, as in Law, legal decisions and the Rule (Lewis[35], Farber[16]), in Game Theory, actions and social norms (Acemoglu and Jackson[2]), in Experimental Economics, decision making and social norms (Vostroknutov[54]), Artificial Intelligence, Reinforcement Learning (Tennenholtz, Merlis, Shani, Mladenov and Boutilier[50]). We also allow to interlink these with many kinds of effects and history-dependencies, and consider a very general instantaneous reward function, general surroundings evolutions and feasible sets.

Thus, our modeling is designed to study decisions and surroundings, simultaneously or not, and to allow the decisions and surroundings to interlink, while keeping it still tractable and interpretable within different contexts. We provide history-dependent models. We obtain optimization results and develop dynamic programming tools to deal with such models.

More precisely, we consider the following three history-dependent frameworks

- a general history-dependent framework (GHDF)
- an easily implementable history-dependent framework (EIHDF)

¹including Ryder and Heal[25], Rozen[44], Rustichini and Siconolfi[46], Caroll, Overland and Weil[10], He, Dyer and Butler[23], Baucells and Sarin[7], Ikefuji[26], Löfgren[36], Safi and Ben Hassen[47].

• a general sustainable framework (GSF)

The first history-dependent model, i.e. the general history-dependent framework (GHDF), is a general tractable framework for intertemporal optimization framework in which the instantaneous reward function depends on the memory of (eventually all) the previous decisions. On one hand, we introduce a general function allowing to model many different memory processes and on the other hand, we deal with general decision variables, objective functions and feasible sets. The formalism we propose allows to discuss and study the memory process formation (through a function m) as well as the way the history enters the instantaneous reward function. We provide dynamic programming tools for such models. Without concavity assumptions, we show the existence of a solution and that the value function is the unique fixed point of the Bellman operator.

Further, we provide an easily implementable history-dependent framework (EIHDF), a version of our general history-dependent framework GHDF which is both easily implementable and as general as needed to be widely applicable. This is allowed by providing a history-dependent framework in which the primer of the problem, in particular the instantaneous reward function and the feasible set Γ , are defined in an adequate recursive way answering both issues simultaneously. We then show that such a model is a particular case of our general history-dependent framework and the results for this particular case are derived as corollaries of the results for the general history-dependent framework.

Finally, taking into account the fact that the environment keeps in memory our activities and decisions, we provide a general sustainable framework (GSF) which introduces a basis for future analysis in environmental and sustainable issues. The GSF model is designed in a very amenable and flexible manner so that it can be adapted to many contexts and one can easily remove or add different effects that will be needed to be addressed. The mathematical results (including existence of a solution and dynamic programming tools) are derived as an application of our general history-dependent framework GHDF and can be directly used.

The GSF encompasses many existing models in the environmental literature, whether linear or circular economy models. Particularly, a whole section of this paper is devoted to applications and examples of GSF to circular economy models. While incorporating the concept of circular economic activities, George Lin and Chen[19] considers a social planner twho maximizes an intertemporal utility function, which depends on aggregate consumption and the stock of pollution. This model was later generalized to incorporate the intensity of recycling by Stengos[30], and to further incorporate recycling habits by Kasioumi[28, 29]. The theoretical circular-economy model of economic growth with circular and cumulative causation (CCC) is presented in Donaghy[13, 14]. The George Lin and Chen[19]'s model is modified by Donaghy[13, 14] by including physical capital K, human capital HC, labor L, and other materials OM, as productive factors. We detail extensively some of these models within our framework and discuss how our framework is fitted for future research. It is amenable not only to treat various sustainable and environmental issues but also allows to connect these with many kinds of effects and history-dependencies (consumption, production, saving and investment, human capital, labor, consumption habits, recycling habits, pollution, stock of waste, etc).

The research on circular and sustainable economy is currently vivid. Our framework is fitted to consider and interlink economic, environmental, technological and social issues. The GSF can easily be adapted to already suggested paths for future research (Donaghy[14], Ghisellini[20], etc) and the extensive literature that is developing. The way the GSF may incorporate the history-dependence viewpoint and the memory formation that we introduce open perspectives towards several aspects and interpretations. In particular, the GSF allows to deal with many important features that are

coming to be taken into account, such as recycling, reuse, reduction, design, habits, activities of harvesting exhaustible and renewable resources, the assimilative capacity of the natural environment for (non-recyclable) waste, transport activities, management of resources, interaction between the processes and the environment, preventative and regenerative eco-industrial development, etc. We discuss the way existing models are particular cases of our general framework as well as how its flexibility allows to use it in future research. As an example, the sustainable process or design variable r does not enter directly Donaghy's utility function in contrast with this possibility which is allowed in our framework. Our general framework allows to study simultaneously many effects and contexts: circular models without production waste (Section 4.4), linear economies with production waste, and furthermore circular economies with production waste, as well as other many effects. These may be interconnected: our general framework allows to enrich the analysis by treating at the same time several kinds of history dependencies. This is crucial for sustainability issues as they involve on one hand habits (consumption habits but also recycling habits), and on another hand pollution stocks and environmental quality (Mazar[38], Moreau[39]). Such a modelling choice also emphasizes the fact that the environment keeps in memory our activities and decisions.

The paper is organized as follows. Section 2 introduces the general history-dependent framework GHDF and the easily implementable history-dependent framework EIHDF, which is a version of our general history-dependent framework GHDF in which the primer of the problem are defined in an adequate recursive way so that it is both kept easily implementable and as general as needed to be widely applicable. We provide the results on the existence of a solution and dynamic programming tools for such models. Section 3 provides the general sustainable framework GSF which introduces a very amenable and flexible basis for future analysis in environmental and sustainable issues and encompasses many existing models in the environmental literature (including linear and circular economy models). We show that it is an application of the GHDF. Section 4 details and addresses many examples and applications of our frameworks from the related literature. The proofs are given in the Appendix. Section 5 concludes.

2 A general history-dependent framework (GHDF)

We propose a general intertemporal optimization framework in which the instantaneous reward function depends on the memory of all the previous decisions. The formalism we propose allows to discuss and study the memory process formation as well as the way the history enters the instantaneous reward function. We develop optimization results and dynamic programming tools. We further provide a general history-dependent framework which is both easily implementable and as general as needed to be widely applicable. This is allowed by providing a history-dependent framework in which the primer of the problem, in particular the instantaneous reward function and the feasible set, are defined in an adequate recursive way answering both issues simultaneously. We then show that such a model is a particular case of our general history-dependent framework and the results for this particular case are derived as corollaries of the results for the general history-dependent framework.

2.1 The general history-dependent model

We now present the general intertemporal optimization framework. The instantaneous reward function depends on the memory of all the previous decisions. We introduce a *memory* function m and discuss the way the history enters the instantaneous reward function (through a function φ). Through these functions, our formalism allows to study many memory process formations and history-dependencies.

We describe a general framework of an infinite horizon intertemporal history dependent decision problem of a social planner which may be a problem of optimal growth related to consumption, saving, conservation, accumulation, pollution or sustainable issues. depending on the objectives and the constraints of the model.

Let X be a topological space which will be the set of state and control variables of the problem. Let us consider an intertemporal decision process. Let Y be a topological space. We define a *memory* function by $m: X \times X \to Y$ such that for time $t, m(x_t, x_{t+1})$ is the *memory* for the decision $x_{t+1} \in X$ given x_t . By this memory function m, we will obtain the history of the memories over time which is modeled in the following way:

Let us consider the set l^{∞}_+ defined by

$$l_{+}^{\infty} = \{ \tilde{x} = (x_t)_{t=0}^{\infty} \in (\mathbb{R}^+)^{\mathbb{N}}, \| \tilde{x} \|_{\infty} := \sup_{t \in \mathbb{N}} x_t < +\infty \}$$

At time t = 0, she has a time-0 history denoted by $\tilde{h}^{(0)} := (h_0^{(0)}, h_1^{(0)}, ...) = (h_j^{(0)})_{j=1}^{\infty}$ lying in l_+^{∞} . Having on memory $m(x_0, x_1)$ at time 0, she has then time-1 history equal to $\tilde{h}^{(1)} := (m(x_0, x_1), \tilde{h}^{(0)})$ at time t = 1. That is, for any time $t \ge 1$, time-t history will be

$$\forall t \ge 1, \tilde{h}^{(t)} = (h_j^{(t)})_{j=1}^{\infty} := (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)})$$

The *j*-th coordinate of the sequence $\tilde{h}^{(t)}$ is denoted by $h_j^{(t)}$. For *j* such that $1 \leq j \leq t$, the term $h_j^{(t)}$ is the decision *j* periods prior to time *t*, i.e.

$$h_{j}^{(t)} = m(x_{t-j}, x_{t-j+1})$$

For $j \ge t+1$, the terms $h_j^{(t)}$ of the sequence $\tilde{h}^{(t)}$ are defined by the terms of $\tilde{h}^{(0)}$, i.e.

$$(h_j^{(t)})_{j=t+1}^{\infty} = \tilde{h}^{(0)}$$

Let us consider an agent facing a time t objective function F which is defined on a subset \mathcal{D}_F of $l^{\infty}_+ \times X \times X$

$$F: \mathcal{D}_F \subseteq (l_+^\infty \times X \times X) \to \mathbb{R}$$

The feasible correspondence $\Gamma: (l^{\infty}_+ \times X) \to X$ is given such that for any $(\tilde{h}, x) \in l^{\infty}_+ \times X$

$$\Gamma(\tilde{h}, x) \subseteq \{ x' \in X, (\tilde{h}, x, x') \in \mathcal{D}_F \}$$

Assuming a fixed discount factor $\beta \in (0, 1)$, for initially given state stock $x_0 > 0$ and time-0 history $\tilde{h}^{(0)} \in l_+^\infty$; with feasible state and control variables at time t, that is, satisfying the process $x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t)$, for every $t \ge 0$, the general framework optimization problem can then be written as

follows:

$$P_{F,\Gamma,m}(\tilde{h}^{(0)}, x_0) = \begin{cases} \max & \sum_{t=0}^{+\infty} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1}) \\ \text{s.t.} & \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t) \\ & \forall t \ge 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}) \\ & x_0 > 0 \text{ and } \tilde{h}^{(0)} \in l_+^{\infty} \text{ are given} \end{cases}$$

2.1.1 Notations and feasible sets

DEFINITION 2.1.— For any given initial data $x_0 > 0$, and initial time-0 history $\tilde{h}^{(0)} \in l_+^{\infty}$, the feasible set $\Pi(k_0, \tilde{h}^{(0)})$ is defined by the set of sequences feasible from x_0 and $\tilde{h}^{(0)}$, i.e. for any $x_0 > 0$, for any $\tilde{h}^{(0)} \in l_+^{\infty}$,

$$\Pi(\tilde{h}^{(0)}, x_0) = \{ \tilde{x} = (x_t)_{t=1}^{+\infty} \in X^{I\!\!N}, \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t), \tilde{h}^{(t+1)} = (m(x_t, x_{t+1}), \tilde{h}^{(t)}) \}$$

The process is analogous as usual. The feasible set is defined from x_0 but here also from the given (infinite) sequence $\tilde{h}^{(0)}$. Once x_1 is chosen in the feasible set $\Gamma(\tilde{h}^{(0)}, x_0)$, the sequence $\tilde{h}^{(1)}$ is updated from the current chosen variable x_1 and the given variable x_0 . Then, x_2 is chosen given x_1 and $\tilde{h}^{(1)}$ in the feasible set $\Gamma(\tilde{h}^{(1)}, x_1)$. And so on, at time t + 1, given x_t and $\tilde{h}^{(t)}$, the variable x_{t+1} must be chosen in the feasible set $\Gamma(\tilde{h}^{(t)}, x_t)$. Once x_{t+1} is chosen, at time t + 2, x_{t+2} is chosen given x_{t+1} and $\tilde{h}^{(t+1)}$ that has been updated from the previous history sequence $\tilde{h}^{(t)}$ and the variables (x_t, x_{t+1}) . Such a process allows to keep decisions in memory while chosing the variable x_t at each time t.

For a sequence $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(\tilde{h}^{(0)}, x_0)$, we denote by $\mathcal{U}(\tilde{x})$ the objective i.e.

$$\mathcal{U}(\tilde{x}) = \sum_{t=0}^{+\infty} \beta^t F(\tilde{h}^{(t)}), x_t, x_{t+1})$$

2.1.2 Assumptions

We set that (F, Γ) satisfies the set (\mathcal{A}) of assumptions if the following assumptions (\mathbf{F}) on the feasible set, (\mathbf{A}) on the instantaneous function, and (\mathbf{M}) on the memory function are satisfied

 (\mathbf{F})

(F1) Γ is a continuous nonempty compact-valued correspondence from $l^{\infty}_{+} \times X$ into X.

(F2) There exists $a \ge 0, a \ne 1$ and $a' \ge 0$ such that for any $x \in X$ and any history $\tilde{h} \in l_+^{\infty}$, $x' \in \Gamma(\tilde{h}, x) \Rightarrow ||x'|| \le a' ||x|| + a$

 (\mathbf{A})

(A1) $\forall x_0 > 0 \text{ and } \tilde{h}^{(0)} \neq 0, \exists \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0) \text{ such that } \mathcal{U}(\tilde{x}) > -\infty.$

(A2) There exist $a \in \mathbb{R}^+$ with $a\beta < 1$, $a_2 \in \mathbb{R}^+$ and a continuous function $a_1 : X \to \mathbb{R}$ such that for any $x_0 > 0$ and $\tilde{h}^{(0)} \in l^{\infty}_+$, for any feasible sequence $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(\tilde{h}^{(0)}, x_0)$ and its associated history $\tilde{h}^{(t)}$, for any $t \ge 0$,

$$F^+(\tilde{h}^{(t)}, x_t, x_{t+1}) \le a_1(x_0)a^t + a_2$$

 (\mathbf{M}) The function m is continuous.

2.1.3 The objective is well-defined

The next result shows that, under the set of assumptions (\mathcal{A}) , the objective function of the optimization problem $\mathcal{P}_{F,\beta,m}(\tilde{h}^{(0)}, x_0)$ is well-defined. The proof is given in the Appendix.

PROPOSITION 2.1.— Assume (A) and let $x_0 \in X$ and $\tilde{h}^{(0)} \in l^{\infty}_+$ be given. Then for any feasible sequence $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(\tilde{h}^{(0)}, x_0)$, the limit $\lim_{T \to +\infty} \sum_{t=0}^{T} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1})$, with $\forall t \geq 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)})$, is well-defined.

The proof can be found in the Appendix.

2.1.4 Existence and uniqueness of the solution

In this section, we give a proposition which shows the existence of a solution to the optimization problem $\mathcal{P}_{F,\beta,m}(\tilde{h}^{(0)}, x_0)$. The uniqueness is obtained under additional assumptions, requiring strict concavity of the instantaneous function F. The proof is given in the Appendix.

PROPOSITION 2.2.— Assume (\mathcal{A}) and assume that $\Pi(\tilde{h}^{(0)}, x_0) \neq \emptyset$. Then there exists an optimal solution. Moreover, if F is jointly strictly concave in (y, x, x') on $l_+^{\infty} \times X \times X$, then the solution is unique.

We give the existence and uniqueness results in a general framework, in particular without any differentiability assumptions. The strict joint concavity of F allows to guarantee uniqueness. The proof can be found in the Appendix.

2.1.5 The value function and Bellman equation

We now define the value function of the optimization problem. We show that under the set of assumptions (\mathcal{A}) , dynamic programming tools can be used to study $\mathcal{P}_{F,\beta,m}(\tilde{h}^{(0)}, x_0)$. We first study the properties of the value function. We then provide an appropriate set of functions on which the Bellman operator has a unique fixed point which is the value function.

DEFINITION 2.2.— The value function V is defined on $l_+^{\infty} \times X$ by for any $(\tilde{h}^{(0)}, x_0) \in l_+^{\infty} \times X$

$$V(\tilde{h}^{(0)}, x_0) = \begin{cases} Max & \sum_{t=0}^{+\infty} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1}) \\ s.t. & \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t) \\ & \forall t \ge 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}) \\ & \tilde{h}^{(0)} \in l_+^{\infty} \text{ and } x_0 \in X \text{ are given} \end{cases}$$

PROPOSITION 2.3.— Assume (A). Then the value function V is upper semi-continuous.

The proof of this proposition can be found in Appendix.

2.1.6 Properties of the value function

The next proposition states some further properties on the value function and the proof is given in Appendix. These properties, together with the upper-semi continuity, will provide an appropriate set of functions to consider for dynamic programming tools.

PROPOSITION 2.4.— Assume (A). Then the value function V satisfies (i) $\forall x_0, \tilde{h}^{(0)}, \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0), \overline{\lim_{t \to +\infty} \beta^t V(\tilde{h}^{(t)}, x_t)} \leq 0.$ (ii) $\forall x_0, \tilde{h}^{(0)}, and \forall \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0)$ such that $\mathcal{U}(\tilde{x}) > -\infty$, $\lim_{t \to +\infty} \beta^t V(\tilde{h}^{(t)}, x_t) = 0.$

The proof can be found in the Appendix.

A standard proof (see Theorem 4.4 p.75 Stokey, Lucas and Prescott[49]) allows to show the following result.

PROPOSITION 2.5. Assume (A). Then \tilde{x}^* is an optimal solution if and only if

$$\forall t \ge 0, V(\tilde{h}^{*(t)}, x_t^*) = F(\tilde{h}^{*(t)}, x_t^*, x_{t+1}^*) + \beta V(\tilde{h}^{*(t+1)}, x_{t+1}^*)$$

where $\tilde{h}^{*(t)} = (m(x_{t-1}^*, x_t^*), m(x_{t-2}^*, x_{t-1}^*), \dots, m(x_1^*, x_2^*), m(x_0^*, x_1^*), \tilde{h}^{(0)})$

Let B be the Bellman operator, i.e. $B: \mathcal{F}(l_+^{\infty} \times X, \mathbb{R}) \to \mathcal{F}(l_+^{\infty} \times X, \mathbb{R})$ be defined by

$$\forall w \in \mathcal{F}(l_+^{\infty} \times X, \mathbb{R}), Bw(\tilde{h}, x) = \max_{x' \in \Gamma(\tilde{h}, x)} \{F(\tilde{h}, x, x') + \beta w((m(x, x'), \tilde{h}), x')\}$$

DEFINITION 2.3.— Let $\mathcal{F}_b(l^{\infty}_+ \times X, I\!\!R)$ be the set of upper semi-continuous functions $w \in \mathcal{F}(l^{\infty}_+ \times X, I\!\!R)$ such that (i) $\forall x_0 \in X, \forall \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0), \lim_{t \to +\infty} \beta^t w(\tilde{h}^{(t)}, x_t) \leq 0,$ with $\tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)})$ (ii) $\forall x_0 \in X, \forall \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0)$ such that $\mathcal{U}(\tilde{x}) > -\infty$, one has $\lim_{t \to +\infty} \beta^t w(\tilde{h}^{(t)}, x_t) = 0$

2.1.7 The value function is the unique fixed-point of the Bellman operator

We now finally state that the value function is the unique fixed-point of the Bellman operator on this set of functions. The proof is given in Appendix.

PROPOSITION 2.6.— Assume (A). Then the value function V is the unique fixed-point of the Bellman operator on the set of functions $\mathcal{F}_b(l^{\infty}_+ \times X, \mathbb{R})$.

Finally, this shows that dynamic programming tools can be used to deal with general historydependent optimal growth models.

2.2 An easily implementable history-dependent model (EIHDF)

In this section, we provide a general history-dependent framework which is both easily implementable and as general as needed to be widely applicable. This is allowed by providing a history-dependent framework in which the primer of the problem, in particular the instantaneous reward function and the feasible set Γ , are defined in an adequate recursive way answering both issues simultaneously. We then show that such a model is a particular case of our general history-dependent framework and the results for this particular case are derived as corollaries of the results for the general history-dependent framework.

2.2.1 The problem

Let X, Y be topological spaces. Let us consider $F : \mathbb{R}^N \times X \times X \to \mathbb{R}$ the instantaneous reward function, $m : X \times X \to Y$ the memory function and the (fixed) discount factor $\beta \in (0,1)$. Let us consider an adjustment level function $\varphi : l_+^{\infty} \to \mathbb{R}^N$, with $N \in \mathbb{N}$. Let $x_0 \in X$ and $\tilde{h}^{(0)} \in l_+^{\infty}$ be given, and consider the problem

$$\mathcal{P}_{F,\Gamma,m}^{\varphi}(\tilde{h}^{(0)}, x_0) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t F(\varphi(\tilde{h}^{(t)}), x_t, x_{t+1}) \\ \text{s.t.} & \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t) \\ & \forall t \ge 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}) \end{cases}$$

Recall that for any given initial data $x_0 > 0$, and initial time-0 history $\tilde{h}^{(0)} \in l_+^{\infty}$, the feasible set $\Pi(\tilde{h}^{(0)}, x_0)$ is defined by the set of sequences feasible from x_0 and $\tilde{h}^{(0)}$, i.e. for any $x_0 > 0$, for any $\tilde{h}^{(0)} \in l_+^{\infty}$,

$$\Pi(\tilde{h}^{(0)}, x_0) = \{ \tilde{x} = (x_t)_{t=1}^{+\infty} \in X^{\mathbb{N}}, \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t), \tilde{h}^{(t+1)} = (m(x_t, x_{t+1}), \tilde{h}^{(t)}) \}$$

2.2.2 Assumptions

Let us give a set (\mathcal{A}') of assumptions: (F) on the feasible set, (A') on the instantaneous function, and (M) on the memory function.

$$(\mathbf{A'})$$

(A1') $\forall x_0 > 0 \text{ and } \tilde{h}^{(0)} \neq 0, \exists \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0) \text{ such that } \sum_{t=0}^{+\infty} \beta^t F(\varphi(\tilde{h}^{(t)}), x_t, x_{t+1})) > -\infty.$

(A2') The function φ is continuous and there exist $a \in \mathbb{R}^+$ with $a\beta < 1$, $a_2 \in \mathbb{R}^+$ and a continuous function $a_1: X \to \mathbb{R}$ such that for any $x_0 > 0$ and $\tilde{h}^{(0)} \in l^{\infty}_+$, for any feasible sequence $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(\tilde{h}^{(0)}, x_0)$ and its associated history $\tilde{h}^{(t)}$, for any $t \ge 0$,

$$F^+(\varphi(h^{(t)}), x_t, x_{t+1}) \le a_1(x_0)a^t + a_2$$

2.2.3 An easily implementable case

An easily implementable case is when the function $\varphi : l^{\infty}_+ \to \mathbb{R}^N$, with $N \in \mathbb{N}$ can be defined recursively through a function $G: Y \times \mathbb{R}^N \to \mathbb{R}^N$, i.e. for any $v \in Y$ and any $\tilde{h} \in l^{\infty}_+$, as

$$\varphi(v,\tilde{h}) = G(v,\varphi(\tilde{h}))$$

and when the correspondence Γ can be defined (and easily calculated) by φ

$$\Gamma(\tilde{h}, x) = g(\varphi(\tilde{h}), x)$$

with $g : \mathbb{R}^N \times X \to X$ is given. The problem is then written

$$\mathcal{P}_{F,\Gamma,m}^{\varphi}(\tilde{h}^{(0)}, x_0) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t F\left(\varphi(\tilde{h}^{(t)}), x_t, x_{t+1}\right) \\ \text{s.t.} & \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t) \\ & \forall t \ge 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}) \\ & \forall t \ge 1, \varphi(\tilde{h}^{(t)}) = G\left(m(x_{t-1}, x_t), \varphi(\tilde{h}^{(t-1)})\right) \\ & x_0 > 0 \text{ and } \tilde{h}^{(0)} \in l_+^{\infty} \text{ are given} \end{cases}$$

2.2.4 A particular case of our general history-dependent framework

It is a particular case of our general history-dependent framework. Let us define the following instantaneous reward function $\hat{F}: l^{\infty}_{+} \times X \times X \to \mathbb{R}$ by

$$\hat{F}(\tilde{h}^{(t)}, x_t, x_{t+1}) = F(\varphi(\tilde{h}^{(t)}), x_t, x_{t+1})$$

It is straightforward to check that the problem $\mathcal{P}_{F,\Gamma,m}^{\varphi}(\tilde{h}^{(0)}, x_0)$ is equivalent to the general historydependent problem $\mathcal{P}_{\hat{F},\Gamma,m}(\tilde{h}^{(0)}, x_0)$ and that (\hat{F}, Γ) satisfies the set (\mathcal{A}) of assumptions (by (\mathcal{A}')),

2.2.5 Theorem/proposition (existence d'une solution, programmation dynamique etc)

The following results are then derived as corollaries of the ones previously shown for the general history-dependent model. Assume (\mathcal{A}') in this subsection hereafter.

COROLLARY 2.1.— Assume (\mathcal{A}') . The problem $\mathcal{P}^{\varphi}_{F,\Gamma,m}(\tilde{h}^{(0)}, x_0)$ is well-defined. Assume moreover that $\Pi(\tilde{h}^{(0)}, x_0) \neq \emptyset$. Then there exists an optimal solution. Moreover, if F is jointly strictly concave in (y, x, x') on $l^{\infty}_+ \times X \times X$ and φ is linear, then the solution is unique.

DEFINITION 2.4.— The value function V is defined on $l_+^{\infty} \times X$ by for any $(\tilde{h}^{(0)}, x_0) \in l_+^{\infty} \times X$

$$V(\tilde{h}^{(0)}, x_0) = \begin{cases} Max & \sum_{t=0}^{+\infty} \beta^t F(\varphi(\tilde{h}^{(t)}), x_t, x_{t+1}) \\ s.t. & \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t) \\ & \forall t \ge 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}) \\ & \tilde{h}^{(0)} \in l_+^{\infty} \text{ and } x_0 \in X \text{ are given} \end{cases}$$

COROLLARY 2.2.— Assume (\mathcal{A}') . Then the value function V is upper semi-continuous.

COROLLARY 2.3.— Assume (\mathcal{A}') . Then the value function V satisfies (i) $\forall x_0, \tilde{h}^{(0)}, \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0), \overline{\lim_{t \to +\infty} \beta^t V(\tilde{h}^{(t)}, x_t)} \leq 0.$ (ii) $\forall x_0, \tilde{h}^{(0)}, and \forall \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0)$ such that $\mathcal{U}(\tilde{x}) > -\infty$, $\lim_{t \to +\infty} \beta^t V(\tilde{h}^{(t)}, x_t) = 0.$

COROLLARY 2.4.— Assume (\mathcal{A}') . Then \tilde{x}^* is an optimal solution if and only if

$$\forall t \ge 0, V(\tilde{h}^{*(t)}, x_t^*) = F(\varphi(\tilde{h}^{*(t)}), x_t^*, x_{t+1}^*) + \beta V(\tilde{h}^{*(t+1)}, x_{t+1}^*)$$

where $\tilde{h}^{*(t)} = (m(x_{t-1}^*, x_t^*), m(x_{t-2}^*, x_{t-1}^*), \dots, m(x_1^*, x_2^*), m(x_0^*, x_1^*), \tilde{h}^{(0)})$

Let B be the Bellman operator, i.e. $B: \mathcal{F}(l_+^{\infty} \times X, \mathbb{R}) \to \mathcal{F}(l_+^{\infty} \times X, \mathbb{R})$ be defined by

$$\forall w \in \mathcal{F}(l_+^{\infty} \times X, \mathbb{R}), Bw(\tilde{h}, x) = \max_{x' \in \Gamma(\tilde{h}, x)} \{ F(\varphi(\tilde{h}), x, x') + \beta w((m(x, x'), \tilde{h}), x') \}$$

$$= \max_{x' \in g(\varphi(\tilde{h}), x)} \{ F(\varphi(\tilde{h}), x, x') + \beta w((m(x, x'), \tilde{h}), x') \}$$

DEFINITION 2.5.— Let $\mathcal{F}_b(l^{\infty}_+ \times X, \mathbb{R})$ be the set of upper semi-continuous functions $w \in \mathcal{F}(l^{\infty}_+ \times X, \mathbb{R})$ such that (i) $\forall x_0 \in X, \forall \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0), \lim_{t \to +\infty} \beta^t w(\tilde{h}^{(t)}, x_t) \leq 0,$ with $\tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)})$ (ii) $\forall x_0 \in X, \forall \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0)$ such that $\mathcal{U}(\tilde{x}) > -\infty$, one has $\lim_{t \to +\infty} \beta^t w(\tilde{h}^{(t)}, x_t) = 0$ COROLLARY 2.5.— Assume (\mathcal{A}') . Then the value function V is the unique fixed-point of the Bellman operator on the set of functions $\mathcal{F}_b(l^{\infty}_+ \times X, \mathbb{R})$.

We next propose a general sustainable framework that provides a basis for future research, including environmental and sustainable analysis. It is a particular case of our general history-dependent framework. This illustrates how our general history-dependent framework can be used to deal with many economic issues.

3 Application: a general sustainable framework (GSF)

In this section, we present a general sustainable framework. It provides a basis for future analysis on environmental and sustainable issues. It is designed in a very flexible manner. It can be adapted to many contexts and one can easily remove or add different effects that are needed to be addressed. The mathematical results (including existence of a solution and dynamic programming tools) can be directly used. They are derived as an application of the general history-dependent framework presented in the previous section.

The GSF encompasses many existing models in the literature. The next section presents the model and the following section is devoted to examples from the literature: we detail extensively some of these models within our framework and discuss how our framework is fitted for future research as it is amenable not only to treat various sustainable and environmental issues but also allows to interlink these with many kinds of effects and history-dependencies (consumption, production, saving and investment, human capital, labor, consumption habits, recycling habits, pollution, stock of waste, social and legal norms, etc).

3.1 The model

We consider an economy in which a unique final good is produced using two kinds of factors of production. The sustainable inputs obtained from the sustainable process (from waste, recycling actions, reuse, etc) are denoted by ξ and the other inputs (as polluting resources, capital, human capital, labor, other materials, etc) are denoted by κ . At each date t, the production level y_t is thus given by the production function \hat{f} depending on² $\xi_t \in \mathbb{R}$ and $\kappa_t \in \mathbb{R}^{N_{\kappa}}$ (with $N_{\kappa} \in \mathbb{N}$) as

$$y_t = \hat{f}(\xi_t, \kappa_t)$$

In this economy, the representative consumer cares for the state of environment and for sustainability. She derives utility from consumption c_t , from the sustainable process or design r_t (as for example the recycling level), and from environmental and sustainability variables E_t (which can be the stock of pollution, the recycling habits, etc). These environmental and sustainability variables depend on all previous decisions-history, thus inducing instantaneous history-dependent preferences. At each date t, the instantaneous utility of the representative agent thus depends on $c_t, r_t \in \mathbb{R}^+$ and $E_t \in \mathbb{R}^{N_E}$ (with $N_E \in \mathbb{N}$) as

$$u(c_t, r_t, E_t)$$

The economy accumulates waste. Indeed, production and consumption generate discards. Discards generated from production come both from the production process itself and the use of the polluting

²Note that $\xi \in \mathbb{R}^N$ ($N \in \mathbb{N}$) can be considered in this framework, it is only for simplicity of exposition that $\xi \in \mathbb{R}$.

input for producing, so the level $\hat{\mathcal{D}}_p$ of discards from production is a function of the production level y_t and the other inputs level κ_t . The level $\hat{\mathcal{D}}_c$ of discards from consumption depends on the consumption level c_t . All what is produced (y_t) and all discards that are neither consumed (c_t) , invested (i_t) nor used in the sustainable process (r_t) , accumulate as waste:

$$s_{t+1} - s_t = \hat{f}(\xi_t, \kappa_t) + \hat{\mathcal{D}}_p(y_t, \kappa_t) + \hat{\mathcal{D}}_c(c_t) - c_t - i_t - r_t$$

with r_t depending on the instantaneous waste stock s_t and on the environmental and sustainability variables E_t . This is modelled through a function \mathcal{R} so that we have $r_t = \mathcal{R}(s_t, E_t)$ such that for all s, E, one has $0 \leq \mathcal{R}(s, E) \leq s$. The investment i_t depends on the instantaneous and the next period input levels, i.e. $i_t = \hat{\mathcal{I}}(\xi_t, \kappa_t, \xi_{t+1}, \kappa_{t+1})$. The waste accumulation dynamics is thus given by

$$s_{t+1} - s_t = \hat{f}(\xi_t, \kappa_t) + \hat{\mathcal{D}}_p(y_t, \kappa_t) + \hat{\mathcal{D}}_c(c_t) - c_t - \hat{\mathcal{I}}(\xi_t, \kappa_t, \xi_{t+1}, \kappa_{t+1}) - \mathcal{R}(s_t, E_t)$$

The environmental and sustainability variables E_t evolve in function of their previous state and the current decisions. The law of motion of the environmental and sustainability variables is thus defined through a given function $G : \mathbb{R} \times \mathbb{R}^{N_{\kappa}} \times \mathbb{R} \times \mathbb{R}^{N_E} \to \mathbb{R}^{N_E}$ by, at each date t,

$$E_{t+1} = G((s_t, \kappa_t, c_t), E_t)$$

Given the initial stock of waste s_0 , available inputs κ_0 , and initial environmental and sustainability variables E_0 , the representative agent solves the following optimization problem

$$\hat{\mathcal{P}} = \begin{cases} \text{Maximize} \sum_{t=0}^{+\infty} \beta^{t} u(c_{t}, r_{t}, E_{t}) \\ \text{s.t. } \forall t \ge 0, \\ s_{t+1} - s_{t} = \hat{f}(\xi_{t}, \kappa_{t}) + \hat{\mathcal{D}}_{p}(y_{t}, \kappa_{t}) + \hat{\mathcal{D}}_{c}(c_{t}) - c_{t} - \hat{\mathcal{I}}(\xi_{t}, \kappa_{t}, \xi_{t+1}, \kappa_{t+1}) - \mathcal{R}(s_{t}, E_{t}) \\ E_{t+1} = G(s_{t}, \kappa_{t}, c_{t}), E_{t}) \\ r_{t} = \mathcal{R}(s_{t}, E_{t}) \in [0, s_{t}] \\ s_{0}, \kappa_{0} > 0, E_{0} \ge 0 \text{ are given} \end{cases}$$

In a sustainable economy, a key assumption is that production involves an input which is obtained through sustainable actions. Here, it is assumed that

$$\xi_t = \mathcal{R}(s_t, E_t)$$

Then, by defining $f(s_t, \kappa_t, E_t) := \hat{f}(\mathcal{R}(s_t, E_t), \kappa_t), \mathcal{D}_p(s_t, \kappa_t) := \hat{\mathcal{D}}_p(f(s_t, \kappa_t), \kappa_t) \text{ and } \mathcal{I}(s_t, \kappa_t, s_{t+1}, \kappa_{t+1}) = \hat{\mathcal{I}}(\mathcal{R}(s_t, E_t), \kappa_t, \mathcal{R}(s_{t+1}, E_{t+1}), \kappa_{t+1}), \text{ the problem is written}$

$$\mathcal{P} = \begin{cases} \text{Maximize} \sum_{t=0}^{+\infty} \beta^t u(c_t, r_t, E_t) \\ \text{s.t. } \forall t \ge 0, \\ s_{t+1} - s_t = f(s_t, \kappa_t, E_t) + \mathcal{D}_p(s_t, \kappa_t) + \mathcal{D}_c(c_t) - c_t - \mathcal{I}(s_t, \kappa_t, s_{t+1}, \kappa_{t+1}) - \mathcal{R}(s_t, E_t) \\ E_{t+1} = G((s_t, \kappa_t, c_t), E_t) \\ r_t = \mathcal{R}(s_t, E_t) \in [0, s_t] \\ s_0, \kappa_0 > 0, E_0 \ge 0 \text{ are given} \end{cases}$$

This framework can be easily adapted to remove or add any environmental and sustainability variables E related effects, as for example the ones from recycling habits, or pollution, etc, adding or removing

them in the preferences and corresponding accumulation laws of motion. For simplicity of exposition, we assumed that ξ and s belong to IR but the framework can easily be adapted to deal with several kinds of sustainable inputs and several kinds of waste that may play different roles in the economy. Linear economy models are encompassed by assuming there is no input coming from any circular process (i.e. all the functions involved are constant with respect to ξ and there is no \mathcal{R} involved). This will be more extensively discussed afterwards.

3.2 Assumptions

Let us assume that the function $c \to c - \mathcal{D}_c(c)$ is bijective and let us define the function $C : \mathbb{R}^{N_v} \to \mathbb{R}$ by for all $(s, \kappa, s', \kappa', E) \in \mathbb{R}^{N_v}$ (with $N_v := 2 + 2N_\kappa + N_E$),

$$C(s,\kappa,s',\kappa',E) = (Id - \mathcal{D}_c)^{-1} (f(s,\kappa,E) + \mathcal{D}_p(s,\kappa) - s' - \mathcal{I}(s,\kappa,s',\kappa') + s - \mathcal{R}(s,E))$$

This gives the consumption level as a function of the stocks of waste, the inputs, and the environmental and sustainability variables.

The general sustainable problem is equivalent to

$$\mathcal{P} = \begin{cases} \text{Maximize} \sum_{t=0}^{+\infty} \beta^{t} u(C(s_{t}, \kappa_{t}, s_{t+1}, \kappa_{t+1}, E_{t}), \mathcal{R}(s_{t}, E_{t}), E_{t}) \\ \text{s.t. } \forall t \ge 0, \\ C(s_{t}, \kappa_{t}, s_{t+1}, \kappa_{t+1}, E_{t}) \ge 0 \\ \forall t \ge 0, E_{t+1} = G((s_{t}, \kappa_{t}, C(s_{t}, \kappa_{t}, s_{t+1}, \kappa_{t+1}, E_{t})), E_{t}) \\ \mathcal{R}(s_{t}, E_{t}) \in [0, s_{t}] \\ s_{0}, \kappa_{0} > 0, E_{0} \ge 0 \text{ are given} \end{cases}$$

Let us consider the following set \mathcal{S} of assumptions:

(S1) The functions $u, f, \mathcal{D}_c, \mathcal{D}_p, \mathcal{I}, \mathcal{R}, G$ are continuous, discards are bounded (both below and from above), and the function $c \to c - \mathcal{D}_c(c)$ is bijective.

(S2) (i) There exists a bounded function $e: \mathbb{R}^{N_{\kappa}+N_Z} \to \mathbb{R}$ such that for all $(s, \kappa, s', \kappa', E) \in \mathbb{R}^{N_v}$,

$$C(s,\kappa,s',\kappa',E) \ge 0 \Rightarrow ||(s',\kappa')|| \le e(s,\kappa,E)$$

(ii) There exists $a \ge 0, a \ne 1$ and $a' \ge 0$ such that for all $(s, \kappa, E) \in \mathbb{R}^{N_{\kappa}+N_E}$,

$$e(s,\kappa,E) \le a' \| (s,\kappa) \| + a$$

(S3) There exist $a \in \mathbb{R}^+$ with $a\beta < 1$, $a_2 \in \mathbb{R}^+$ and a continuous function $a_1 : X \to \mathbb{R}$ such that for any $x_0 = (s_0, \kappa_0) > 0$ and E_0 , for any feasible sequence $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(E_0, x_0)$ and its associated history $\tilde{h}^{(t)}$, for any $t \ge 0$,

$$u^+(C(s_t,\kappa_t,s_{t+1},\kappa_{t+1},E_t),\mathcal{R}(s_t,E_t),E_t) \le a_1(x_0)a^t + a_2$$

These assumptions are usual (see Le Van Dana[52]). They are not quite restrictive (see Le Van and Morhaim[33]). They allow to cover situations with unboundedness in the objective function combined with various types of the feasible set as used in the literature (with various returns to scale technology).

They ensure that the assumptions in Section 2.2.2 are satisfied. The assumption that $(Id - D_c)$ is bijective allows to uniquely express the instantaneous consumption in terms of the waste stocks, the other inputs stocks, and environmental and sustainable variables, and greatly simplifies the notations. The continuity in (S1) and (S2) (i) ensure (F1) so that Γ is compact-valued. The existence of a function e in Assumption (S2) (ii) ensures Assumption (F2) and Assumptions (S3) and (S4) ensure Assumption (A2').

3.3 The general sustainable framework GSF is a particular case of the general history-dependent framework GHDF

In this section, we show that the GSF is a particular case of the GHDF. Recall that the GSF is

$$\mathcal{P} = \begin{cases} \text{Maximize} \sum_{t=0}^{+\infty} \beta^{t} u(C(s_{t}, \kappa_{t}, s_{t+1}, \kappa_{t+1}, E_{t}, \mathcal{R}(s_{t}, E_{t}), E_{t}) \\ \text{s.t. } \forall t \ge 0, \\ C(s_{t}, \kappa_{t}, s_{t+1}, \kappa_{t+1}, E_{t}) \ge 0 \\ E_{t+1} = G((s_{t}, \kappa_{t}, C(s_{t}, \kappa_{t}, s_{t+1}, \kappa_{t+1}, E_{t})), E_{t}) \\ \mathcal{R}(s_{t}, E_{t}) \in [0, s_{t}] \\ s_{0}, \kappa_{0} > 0, E_{0} \ge 0 \text{ are given} \end{cases}$$

We explain hereafter that the problem is indeed a particular case of our GHDF. Let us define $X := \mathbb{R}^{1+N_{\kappa}}$ and $x := (s, \kappa)$. Let the memory function $m : X \times X \to Y$ be defined with $Y := \mathbb{R}^{2+2N_{\kappa}}$ by for any $(x, x') \in X \times X$,

$$n(x, x') = (x, x')$$

Let us define the function \hat{G} by for any $(x, x') \in X \times X, E \in \mathbb{R}^{N_E}$,

$$\hat{G}((x, x'), E) = G((x, C(x, x', E)), E)$$

The adjustment level function $\varphi: l^{\infty}_{+} \to Z$ is defined recursively in the following way:

$$\forall \tilde{h} \in l^{\infty}_{+}, \forall (x, x') \in X \times X, \varphi(m(x, x'), \tilde{h}) = \hat{G}(m(x, x'), \varphi(\tilde{h}))$$

By defining the instantaneous function F for all $\tilde{h} \in l_+^{\infty}$ and $x, x' \in \mathbb{R}^2$,

$$F(\tilde{h}, x, x') = u(C(x, x', \varphi(\tilde{h})), \mathcal{R}(\pi_1(x), \varphi(\tilde{h})), \varphi(\tilde{h}))$$

and the feasible correspondence Γ ,

$$\Gamma(\tilde{h}, x) = \{ x' \in (\mathbb{R}^+)^2, C(x, x', \varphi(\tilde{h})) \ge 0 \}$$

that can also be written

$$\Gamma(\tilde{h}, x) = g(\varphi(\tilde{h}), x)$$

with the correspondence $g: Z \times X \to X$ defined for all $(k, x) \in Z \times X$ by

$$g(k, x) = \{x' \in \mathbb{R}^2, C(x, x', k) \ge 0\}$$

Then the general sustainable problem is written

$$P(\tilde{h}^{(0)}, x_0) = \begin{cases} \text{Max} & \sum_{t=0}^{+\infty} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1}) \\ \text{s.t.} & \forall t \ge 0, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t) \\ & \forall t \ge 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}) \\ & x_0 > 0 \text{ and } \tilde{h}^{(0)} \in l_+^{\infty} \text{ are given} \end{cases}$$

which shows that our GSF fits GHDF. Hence, the results shown in Section 2 can be applied to the general sustainable problem. In particular, we obtain the existence of a solution and the dynamic programming tools described in Section 2 can be implemented.

This general sustainable framework can be easily adapted to various contexts and models. In the next section, we give existing models in the related literature as examples of our framework.

4 Some models and examples

In this section, we discuss how our general framework allows to study issues and models from various literature strands. As we introduced a general function allowing to model many different memory processes, general decision variables, objective functions and feasible sets, we are able to encompass many existing models. It generalizes the history-dependent intertemporal optimization models provided in Morhaim and Ulus[40]. Thus, the applications presented in Morhaim and Ulus[40] are encompassed. These include seminal models dealing with habit formation (Ryder and Heal[25], Rozen[44], Rustichini and Siconolfi[46], Caroll, Overland and Weil[10]) and satiation (He, Dyer and Butler[23], Baucells and Sarin[7]). Moreover, it encompasses environmental models (Ikefuji[26], Löfgren[36]) and Safi and Ben Hassen[47]) and optimal management of natural resources (Smulders, Toman and Withagen[48], Ulus[51]), as well as recent circular economy models (George, Chi-ang Lin and Chen[19], Kasioumi[28], Kasioumi and Stengos[31]) and circular and causation models (Donaghy[13, 14]).

We next discuss the way these existing models are particular cases of our general framework as well as how its flexibility allows to use it in future research. Our general framework indeed allows to study simultaneously many effects and contexts: circular models without production waste (Section 4.4), linear economies with production waste, and furthermore circular economies with production waste, as well as other many effects.

4.1 History-dependent optimal growth models with (consumption) habit formation or satiation

In Morhaim and Ulus[40], a representative agent consumes a single good on periods t = 0, 1, 2, ...and maximizes her intertemporal utility over the consumption stream $\tilde{c} = (c_0, c_1, ...)$ in l^{∞} . At date t, the consumer's instantaneous utility depends on her current consumption c_t . But it also depends on her *time-t* (consumption) history $\tilde{h}^{(t)}$ which is defined from an initial history $\tilde{h}^{(0)} \in l_+^{\infty}$ and keeps in memory the consumption decisions as follows:

$$\forall t \ge 1, \tilde{h}^{(t)} = (h_j^{(t)})_{j=1}^{\infty} := (c_{t-1}, \tilde{h}^{(t-1)}) = (c_{t-1}, \dots, c_0, \tilde{h}^{(0)})$$

Her time t-utility $u : \mathcal{D}_u \subseteq (\mathbb{R}^+ \times R) \to \mathbb{R} \cup \{-\infty\}$ changes endogenously from her time-t consumption history $\tilde{h}^{(t)}$ through the adjustment level function $\varphi : l^{\infty}_+ \to R$ with $R = (\mathbb{R}^+)^n$ (where $n \geq 1$). For initially given capital stock $k_0 > 0$ and time-0 history $\tilde{h}^{(0)} \in l^{\infty}_+$, the general framework and optimization problem, with $f : \mathbb{R}^+ \to \mathbb{R}^+$ the production function, k_t the capital stock at time t, and $\beta \in (0, 1)$ the fixed discount factor, is given as follows:

$$\mathcal{P}_{u,\varphi,\beta}(k_0, \tilde{h}^{(0)}) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t u(c_t, \varphi(\tilde{h}^{(t)})) \\ \text{s.t.} & \forall t \ge 0, k_{t+1} = f(k_t) - c_t, k_t \ge 0 \text{ and } c_t \ge 0 \\ & \forall t \ge 1, \tilde{h}^{(t)} = (c_{t-1}, \dots, c_1, c_0, \tilde{h}^{(0)}) \\ & \forall t \ge 0, (c_t, \varphi(\tilde{h}^{(t)})) \in \mathcal{D}_u \subseteq (\mathbb{R}^+ \times R) \end{cases}$$

The problem can be rewritten as follows:

$$\mathcal{P}_{u,\varphi,\beta}(k_0,\tilde{h}^{(0)}) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t u(f(k_t) - k_{t+1},\varphi(\tilde{h}^{(t)})) \\ \text{s.t.} & \forall t \ge 0, k_{t+1} \in [0, f(k_t)] \\ & \forall t \ge 1, \tilde{h}^{(t)} = (f(k_{t-1}) - k_t, f(k_{t-2}) - k_{t-1}, \dots, f(k_1) - k_2, f(k_0) - k_1, \tilde{h}^{(0)}) \\ & \forall t \ge 0, (f(k_t) - k_{t+1}, \varphi(\tilde{h}^{(t)})) \in \mathcal{D}_u \subseteq (\mathbb{R}^+ \times R) \end{cases}$$

This is a case in which *consumption* is kept in memory, which is a function of previous date and current capital stock decisions. Thus, it means keeping in memory a function of these decisions. Our general history-dependent framework allows, not only to keep in memory the particular function defining consumption, but any function of the previous date and current decisions. By this way, the model provided in Morhaim and Ulus[40] as well as the models presented in Morhaim and Ulus[40]³ become particular cases of the GHDF. Indeed, let us define $X = Y = \mathbb{R}^+$, $x_t = k_t \in X$ for all $t = 0, \ldots, +\infty$, and $F : l_+^{\infty} \times X \times X \to \mathbb{R}$ by for any $\tilde{h} \in l_+^{\infty}$ and $x, x' \in X$,

$$F(\tilde{h}, x, x') := u(f(x) - x', \varphi(\tilde{h}))$$

We define for any decision x' given x, the memory function/process $m: X \times X \to Y$ by

$$m(x, x') = f(x) - x'$$

and for history $\tilde{h} \in l^{\infty}_{+}$ and for any $x \in \mathbb{R}^{+}$, the feasible correspondence Γ is given by

$$\Gamma(\tilde{h}, x) = \{ x' \in [0, f(x)], (f(x) - x', \varphi(\tilde{h})) \in \mathcal{D}_u \}$$

The introduction of the memory function m to the modelling allows to explicitly study many memory processes. Here, it is done by defining m which associates the consumption to the decisions.

4.2 History-dependent optimization models with environmental effects

History-dependence is important in environmental economics models. Morhaim and Ulus[40] already underlined that their general framework allows to deal with environmental effects. We show in this section how Löfgren[36]'s model is written through our new GHDF. Further, we describe how the discrete time version of Ikefuji[26]'s model is also encompassed by GHDF.

Löfgren[36] proposes a model with environmental quality habit formation and in which a consumption good moreover causes a negative external effect on the environment. The social planner maximizes the

³including Ryder and Heal[25], Rozen[44], Rustichini and Siconolfi[46], Caroll, Overland and Weil[10], He, Dyer and Butler[23], Baucells and Sarin[7], Ikefuji[26], Löfgren[36], Safi and Ben Hassen[47].

utility given the negative effect of the consumption good on the environment and taking into account that there is habit formation in environmental quality. The instantaneous utility $u(n_t, x_t, z_t, s_t)$ depends on n_t which is the environment that displays habit formation, x_t the "dirty" consumption good (the environmental bad), z_t the "clean" consumption good and s_t the habit level related to the environment. The following relations are satisfied with $\gamma \in (0, 1), \beta \in (0, 1), \delta \in (0, 1), y$ being an exogenously given income and n is a given initial environment

$$\begin{cases} n_t = n - \gamma x_t \\ z_t = y - x_t \\ s_{t+1} = \beta n_t + (1 - \delta) s_t \end{cases}$$

Let us define $X = Y = \mathbb{R}^+ x_t = k_t \in X$ for all $t = 0 \dots \infty$, and φ is defined through the following recurrence relation

$$\forall c \in \mathrm{I\!R}^+, \forall \tilde{h} \in l^\infty_+, \varphi((c, \tilde{h})) = \beta n + (1 - \delta)\varphi(\tilde{h}) - \beta \gamma c$$

Let us define $F: l_+^{\infty} \times X \times X \to \mathbb{R}$ by, for any $\tilde{h} \in l_+^{\infty}$ and $x, x' \in X$,

$$F(\tilde{h}, x, x') := u(n - \gamma(f(x) - x'), f(x) - x', y - (f(x) - x'), \varphi(\tilde{h}))$$

We define for any decision x' given x, the memory function/process $m: X \times X \to Y$ by m(x, x') = f(x) - x', and for any $\tilde{h} \in l^{\infty}_+$ and $x \in \mathbb{R}^+$, the feasible correspondence Γ is given by

$$\Gamma(\tilde{h}, x) = \{ x' \in [0, f(x)], (n - \gamma(f(x) - x'), f(x) - x', y - (f(x) - x'), \varphi(\tilde{h})) \in \mathcal{D}_{\tilde{u}} \}$$

Thus, Löfgren[36]'s model is encompassed in our general framework.

Ikefuji[26] studies habit formation in consumption and pollution abatement activities when agents derive disutility both from the habit stock and pollution. The pollution P_t in period t is generated by the capital stock k_t used in production and reduced by abatement activities a_t in the same period. The problem is written as follows.

$$P(h_0, k_0, m_0) = \begin{cases} \text{Max} & \sum_{t=0}^{+\infty} \beta^t u(c_t, H_t, P_t) \\ \text{s.t.} & \forall t \ge 1, H_{t+1} = \rho c_t + (1-\rho) H_t \text{ and } P_t = \left(\frac{k_t}{a_t}\right)^{\phi} \\ & \forall t \ge 0, k_{t+1} = Ak_t - c_t - a_t - k_t \\ & k_0 > 0, m_0 > 0 \text{ and } h_0 \text{ are given} \end{cases}$$

where c_t denotes the consumption in period t, H_t denotes the consumption habit, and P_t is the level of aggregated pollution in the economy.

In order to cover this problem in our general framework, let us define $X = (\mathbb{R}^+)^2$, $Y = \mathbb{R}^+$, and $x_t = (k_t, a_t) \in X$ for all $t = 0 \dots \infty$. The function $\varphi : l_+^{\infty} \to \mathbb{R}$ is defined by $\varphi(\tilde{h}^{(0)}) = h_0$ and

$$\forall h \in l^{\infty}_{+}, \forall c \in \mathbb{R}_{+}, \varphi(c, h) = G(c, \varphi(h)).$$

with the function G defined by $G(c, y) = \rho c + (1 - \rho)y$. Also, let us define $F : l_+^{\infty} \times X \times X \to \mathbb{R}$ by, for any $\tilde{h} \in l_+^{\infty}$ and $x, x' \in X$,

$$F(\tilde{h}, x, x') := u(f(\pi_1(x)) - \pi_1(x'), \varphi(\tilde{h}), (\frac{\pi_1(x)}{\pi_2(x)})^{\phi})$$

We define for any decision x' given x, the memory function/process $m: X \times X \to Y$ by

$$m(x, x') = f(\pi_1(x)) - \pi_1(x')$$

and for history $\tilde{h} \in l^{\infty}_+$ and for any $x \in \mathbb{R}^+$, the feasible correspondence Γ is given by

$$\Gamma(\tilde{h}, x) = \{ x' \in [0, f(\pi_1(x))], \pi_2(x') > 0, \left(f(\pi_1(x)) - \pi_1(x'), \varphi(\tilde{h}), \left(\frac{\pi_1(x)}{\pi_2(x)}\right)^{\phi} \right) \in \mathcal{D}_u \},\$$

showing that our GHDF fits Ikefuji[26]'s model.

These two examples illustrate how our framework is general and easy to use, while keeping the environmental effects to be studied visible and interpretable. In the next section, we show how it is also fitted to optimal management of natural resources.

4.3 Optimal management of natural resources

Smulders, Toman and Withagen[48] and Ulus[51] present models referring to the optimal management of exhaustible natural resources like oil, coal, gas and etc. In these models, there exists a single planner who manages the decision of extraction and consumption of an exhaustible natural resource at each period t in order to maximize the intertemporal utility. At each period t, the instantaneous utility depends on its consumption c_t and also on the stock of the natural resource, which is denoted by s_t . The output $y_t \in \mathbb{R}_+$ is produced from capital (k_t) and extracted quantity of resource (r_t) (or the extraction flow of natural resource per unit of time) by a production function $f : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}_+$ where $y_t = f(k_t, r_t)$. The output is either consumed as $c_t \ge 0$ or saved as capital to the next period as k_{t+1} satisfying:

$$c_t + k_{t+1} \le f(k_t, r_t) + (1 - \delta)k_t$$
 with $k_t \ge 0$

where δ stands for the depreciation rate of capital. For initially given stocks of $k_0, s_0, r_0 > 0$ and $\beta \in (0, 1)$ the fixed discount factor, the problem of the planner is given as follows:

$$\mathcal{P}((k_0, r_0), s_0)) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t u(c_t, s_t) \\ \text{s.t.} & \forall t \ge 0, s_{t+1} = s_t - r_t \\ & \forall t \ge 0, 0 \le c_t = f(k_t, r_t) + (1 - \delta)k_t - k_{t+1} \\ & s_0, k_0, r_0 > 0 \text{ are given} \end{cases}$$

which can be rewritten as

$$\mathcal{P}((k_0, r_0), s_0)) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t u(f(k_t, r_t) + (1 - \delta)k_t - k_{t+1}, s_t) \\ \text{s.t.} & \forall t \ge 0, s_{t+1} = s_t - r_t \\ & \forall t \ge 0, 0 \le k_{t+1} \le f(k_t, r_t) + (1 - \delta)k_t \\ & s_0, k_0, r_0 > 0 \text{ are given} \end{cases}$$

Let us define G by G(s, r) := s - r and φ by the law of motion

$$\forall t \ge 0, \varphi(\tilde{h}^{(t+1)}) = G(r_t, \varphi(\tilde{h}^{(t)}))$$

so that the instantaneous utility $u(c_t, s_t)$ is $u(c_t, \varphi(\tilde{h}^{(t)}))$. By defining $\tilde{h}^{(0)}$ such that $\varphi(\tilde{h}^{(0)}) = s_0$ and for any $t \ge 0$, $\tilde{h}^{(t+1)}$ such that $\tilde{h}^{(t+1)} = (r_t, \tilde{h}^{(t)})$, the model can be written

$$\mathcal{P}_{u,\varphi,\beta}(k_0, \tilde{h}^{(0)}) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t u(f(k_t, r_t) + (1-\delta)k_t - k_{t+1}, \varphi(\tilde{h}^{(t)})) \\ \text{s.t.} & \forall t \ge 0, k_{t+1} \in [0, f(k_t, r_t) + (1-\delta)k_t] \\ & \forall t \ge 0, \tilde{h}^{(t+1)} = (r_t, \dots, \tilde{h}^{(0)}) \end{cases}$$

The model can be written in our general framework by defining x := (k, r) and the memory function by $m(x, x') = \pi_2(x)$. Let us define the objective function F by, for any (x, x', \tilde{h}) ,

$$F(x, x', \tilde{h}) := u \big(f(\pi_1(x), \pi_2(x)) + (1 - \delta)\pi_1(x) - \pi_1(x'), \varphi(\tilde{h}) \big)$$

and the feasible set by, for any (\tilde{h}, x) ,

$$\Gamma(\tilde{h}, x) := \{ x', 0 \le \pi_1(x') \le f(\pi_1(x), \pi_2(x)) + (1 - \delta)\pi_1(x), 0 \le \pi_2(x') \le \varphi(\tilde{h}) \}$$

Here G is defined by G(s, r) := s - r. However, our framework allows to consider more general forms of G, for example if it comes to taking into account the resources regeneration (whether naturally or otherwise).

In the next section, we show that our results are also fitted for circular economy issues.

4.4 Circular economy (CE) and circular and cumulative causation (CCC) models

Circular economy is most frequently depicted as a combination of reduce, reuse and recycle activities (Kirchherr, Reike and Hekkert[32]), but also design, implying a focus on the entire life cycle of the processes as well as the interaction between the process and the environment and the economy in which it is embedded (Ghisellini, Cialani and Ulgiati[20]).

In this section, we show that the circular economy models provided by George, Chi-ang Lin and Chen[19], Kasioumi and Stengos[30], Kasioumi[29] and Donaghy[13, 14] are particular cases of our general sustainable framework. We first consider circular economy models without recycling habits (George, Chi-ang Lin and Chen[19] and Kasioumi and Stengos[30]), then circular economy models with recycling habits (Kasioumi[28]). We also show that our framework is adapted to the circular and cumulative causation models as developed by Donaghy[13, 14].

4.4.1 CE models without recycling habits

In this section, we consider the circular models without recycling habits proposed by George, Chi-ang Lin and Chen[19], and Kasioumi and Stengos[30].

George, Chi-ang Lin and Chen[19] and Kasioumi and Stengos[30] consider a closed economy with zero population growth. They abstract from capital accumulation and technical progress⁴. The social planner maximizes an intertemporal utility where the instantaneous utility u(c, P) depends on consumption c and the stock of pollution P. The output is produced via a (concave) production function ϕ using two factors of production, one which corresponds to the rate of use of the recyclable

⁴However, our results apply to such models with capital accumulation and technical progress (see Section 4.4.3).

resource and another (z) corresponding to the rate of use of the environmentally polluting resource. Output produced in any given period but not consumed or used for the employment of the polluting resource, accumulates as (potentially recyclable) waste. Recycling turns waste into a useful factor of production: a proportion b of the waste stock s with intensity of recycling τ is supposed to be recycled each period. George, Chi-ang Lin and Chen[19]'s model is a particular case of Kasioumi and Stengos[30] in which the intensity of recycling τ is equal to one. The social planner solves the following optimization problem

$$\mathcal{P}((s_0, z_0), p_0) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t u(c_t, p_t) \\ \text{s.t.} & \forall t \ge 0, s_{t+1} - s_t = \phi(\tau b s_t, z_t) - c_t - \alpha z_{t+1} - \tau b s_t \\ & \forall t \ge 0, p_{t+1} - p_t = \theta z_t - \delta p_t + (1-b) s_t \\ & s_0, z_0 > 0, p_0 \ge 0 \text{ are given} \end{cases}$$

By defining $\kappa := z$ (thus $N_{\kappa} = 1$), E := p (thus $N_E = 1$), $f(s, \kappa, E) := \phi(\tau b s, \kappa)$, $\mathcal{D}_p(s, \kappa) := 0$, $\mathcal{D}_c(c) := 0$, $\mathcal{I}(s, \kappa, s', \kappa') := \alpha \kappa'$, $\mathcal{R}(s, E) := \tau b s$ and $G((s, \kappa, c), E) := (1 - \delta)E + \theta \kappa + (1 - b)s$, both models (George, Chi-ang Lin and Chen[19] and Kasioumi and Stengos[30])) fit our general sustainable framework.

4.4.2 CE models with recycling habits

In this section, we consider the circular models with recycling habits proposed by Kasioumi[28, 29]. It is an extension of the theoretical work of Kasioumi and Stengos[30], combining elements of the circular economy model of George, Chi-ang Lin and Chen[19] with the habit formation framework of Ikefuji[26]. Kasioumi[28, 29] deals with a fixed intensity of recycling while Kasioumi[29] deals further with an intensity $\tau(H_t)$ depending (through an affine function $\tau(H_t) := \eta + rH_t$) on the level of recycling habits. The social planner solves the following optimization problem

$$\mathcal{P}((s_0, z_0), p_0) = \begin{cases} \text{Maximize} & \sum_{t=0}^{+\infty} \beta^t u(c_t, r_t, H_t, p_t) \\ \text{s.t.} & \forall t \ge 0, s_{t+1} - s_t = \phi(\tau(H_t) bs_t, z_t) - c_t - \alpha z_{t+1} - \tau(H_t) bs_t \\ & \forall t \ge 0, p_{t+1} - p_t = \theta z_t - \delta p_t + (1 - b)s_t + \rho c_t \\ & H_{t+1} - H_t = \mu(\tau(H_t) bs_t - H_t) \\ & s_0, z_0 > 0, p_0 \ge 0 \text{ are given} \end{cases}$$

By defining $\kappa := z$ (thus $N_{\kappa} = 1$), E := (p, H) (thus $N_E = 2$), $f(s, \kappa, E) := \phi(\tau(\pi_2(E))bs, e)$, $\mathcal{D}_p(s, \kappa) := 0$, $\mathcal{D}_c(c) := 0$, $\mathcal{I}(s, \kappa, s', \kappa') := \alpha \kappa'$, $\mathcal{R}(s, E) := \tau(\pi_2(E))bs$ and

$$G((s,\kappa,c),E) := ((1-\delta)\pi_1(E) + \theta\kappa + (1-b)s + \rho c, \mu(\tau(\pi_2(E))bs - \pi_2(E)))$$

these models (Kasioumi[28, 29]) fit our general sustainable framework. Note that our framework fits also recycling intensity functions τ that are not necessarily affine.

4.4.3 CCC models

The theoretical circular-economy model of economic growth with circular and cumulative causation (CCC) is presented in Donaghy[13, 14] as follows. The George Lin and Chen[19]'s model is modified by modelling capital formation and technical change and including physical capital K, human capital

HC, labor L, and other materials OM, as productive factors. The optimization problem for the social planner is to choose control variables, consumption c and environmentally polluting resource z but also invesments in physical and social capital to maximize the intertemporal utility function.

Given the functions $b : \mathbb{R} \to \mathbb{R}$ accounting for increased efficiencies in recycling (thus depends on the accumulated recycling experience $R \in \mathbb{R}$), the function $\vartheta : \mathbb{R}^2 \to \mathbb{R}$ characterizing technical progress in reducing pollution from the polluting resource as a function of human capital deepening (thus depends on $(K, HC) \in \mathbb{R}^2$), the production function ϕ , and the instantaneous utility function u depending on consumption and pollution $(c, P) \in \mathbb{R}^2$, the model in Donaghy[13, 14] is as follows.

$$\mathcal{P} = \begin{cases} \text{Maximize} \sum_{t=0}^{+\infty} \beta^{t} u(c_{t}, P_{t}) \\ \text{s.t. } \forall t \geq 0, \\ s_{t+1} - s_{t} = \phi(s_{t}, z_{t}, K_{t}, HC_{t}, L_{t}, OM_{t}, R_{t}) - c_{t} - \alpha z_{t+1} \\ -(K_{t+1} - K_{t})(1 + \frac{a_{1}}{2K_{t}}) - (HC_{t+1} - HC_{t})(1 + \frac{a_{2}}{2HC_{t}}) - b(R_{t})s_{t} \\ P_{t+1} - P_{t} = \vartheta(K_{t}, HC_{t})z_{t} - \delta P_{t} + (1 - b(R_{t}))s_{t} \\ R_{t+1} - R_{t} = b(R_{t})s_{t} \\ s_{0}, z_{0} > 0, P_{0} \geq 0 \text{ are given} \\ K_{0}, R_{0}, HC_{0}, L_{0} \geq 0 \text{ are given} \end{cases}$$

Let us define $\kappa := (z, K, L, HC, OM)$ (thus $N_{\kappa} = 5$), E := (P, R), $\mathcal{I}(s, \kappa, s', \kappa') := \alpha \pi_1(\kappa') + (\pi_2(\kappa') - \pi_2(\kappa))(1 + \frac{a_1}{2\pi_2(\kappa)}) + (\pi_4(\kappa') - \pi_4(\kappa))(1 + \frac{a_2}{2\pi_4(\kappa)})$ with for any $i = 1, ..., 5, \pi_i$ is the *i*-th projection⁵, $\mathcal{D}_p(s, \kappa) = 0, \mathcal{D}_c(c) = 0$ and $\mathcal{R}(s, E) = b(\pi_2(E))s$ and G by

$$G((s,\kappa,c),E) = \left(\vartheta(\pi_2(\kappa),\pi_4(\kappa))\pi_1(\kappa) + (1-\delta)\pi_1(E) + (1-b(\pi_2(E)))s,\pi_2(E) + b(\pi_2(E))s\right)$$

The sustainable process or design variable r does not enter directly Donaghy's utility function in contrast with this possibility which is allowed in our framework. When the process does not enter directly in the utility, it suffices to define the utility function $\hat{u}(c, r, M) := u(c, M)$. This shows that circular and cumulative causation models fit our GSF. Note that Donaghy[13] considers the particular functions $b(R) = \left(\frac{\phi}{R} + \zeta\right)^{-1}$, $\vartheta(K, HC) = \left[g - \chi\left(\frac{HC}{K}\right)^{\epsilon}\right]$, $u(c, P) = \left[\frac{1}{\gamma}\left(\omega cP^{-\eta}\right)^{\gamma}\right]$ and $\phi(s, z, K, HC, L, OM, R) = \psi[\theta_1(b(R)s)^{-\sigma} + \theta_2 z^{-\sigma} + \theta_3 K^{-\sigma} + \theta_4 HC^{-\sigma} + \theta_5 L^{-\sigma} + \theta_6 OM^{-\sigma}]^{-\frac{\kappa}{\sigma}}$, whereas we allow general functions.

4.4.4 On the circular and sustainable economy research agenda

The research on circular and sustainable economy is currently vivid. Our framework is fitted to consider and interlink economic, environmental, technological and social issues. The GSF can easily be adapted to already suggested paths for future research (Donaghy[13, 14], Ghisellini[20]) and the extensive literature that is developing. The way the GSF may incorporate the history-dependence viewpoint, and the memory formation that we introduce, open perspectives towards several aspects and interpretations. In particular, as a by-product, the GSF allows to deal with many important features that are coming to be taken into account, such as recycling, reuse, reduction, design, habits, activities of harvesting exhaustible and renewable resources, the assimilative capacity of the natural environment for (non-recyclable) waste, transport activities, management of resources, interaction

⁵i.e. $\pi_i : \mathbb{R}^5 \to \mathbb{R}$ is defined by, for any $a = (a_j)_{j=1}^5 \in \mathbb{R}^5, \pi_i(a) = a_i$

between the processes and the environment, preventative and regenerative eco-industrial development, etc. These may be interconnected.

Our framework is fitted for future research as it is amenable not only to treat various sustainable and environmental issues but also allows to interlink these with many kinds of effects and historydependencies (consumption, production, saving and investment, human capital, labor, consumption habits, recycling habits, pollution, stock of waste, social and legal norms). The way the GSF may incorporate the history-dependence viewpoint and the memory formation that we introduce open perspectives towards several aspects and interpretations. In particular, the GSF allows to deal with many important features that are coming to be taken into account, such as recycling, reuse, reduction, design, habits, activities of harvesting exhaustible and renewable resources, the assimilative capacity of the natural environment for (non-recyclable) waste, transport activities, management of resources, interaction between the processes and the environment, preventative and regenerative eco-industrial development, etc. We discuss the way existing models are particular cases of our general framework as well as how its flexibility allows to use it in future research. As an example, the sustainable process or design variable r does not enter directly Donaghy's utility function in contrast with this possibility which is allowed in our framework. The GSF allows to study simultaneously many effects and contexts: circular models without production waste, linear economies with production waste, and furthermore circular economies with production waste, as well as other many effects. These may be interconnected: our general framework allows to enrich the analysis by treating at the same time several kinds of history dependencies. This is crucial for sustainability issues as they involve on one hand habits (consumption habits but also recycling habits), and on another hand pollution stocks and environmental quality (Mazar[38], Moreau[39]). Such a modelling choice also emphasizes the fact that the environment keeps in memory our activities and decisions.

Following the research program suggested by Donaghy[13, 14], our modelling is a step towards a clearer view on interlinked economic activities and surroundings, which are crucial when dealing with environmental issues and circular economies, thus helping in managing transitions in the economy, and in particular transitions to a circular economy (or system of circular economies).

5 Conclusion

In this paper, we introduced a new framework for the analysis of decisions' processes with historydependencies and surroundings. The modeling we provide is very general and flexible while we keep the framework being interpretable and tractable. It is designed to be adapted to various surroundings and different kinds of history-dependencies. It allows to study the history-dependencies formation, and to highlight how history dependencies affect our decisions as well as how they are affected by our decisions.

We have developed dynamic programming tools (in particular, existence of a solution and that the value function is the unique fixed point of the Bellman operator) to solve such models.

Since environmental and sustainable variables are influenced by (the memory of our past) decisions and can be taken as surroundings, as a by-product, we introduced a very general sustainable framework which fits many existing environmental and sustainable models including circular economy models. It provides a basis for future environmental analysis (as we discussed in Section 4.4.4).

Our framework opens new paths towards the interdependencies between surroundings and decisions. It is amenable not only to treat various issues and phenomena from diverse areas, as in Law, legal decisions and the Rule (Lewis[35], Farber[16]), dynamic models of Law ([8]), in Game Theory, actions and social norms (Acemoglu and Jackson[2]), in Experimental Economics, decision making and social norms (Vostroknutov[54]), Artificial Intelligence, Reinforcement Learning (Tennenholtz, Merlis, Shani, Mladenov and Boutilier[50]). It also allows to interlink these with many kinds of effects and history-dependencies.

6 Appendix

6.1 Proof of Proposition 2.1

By the assumption (\mathcal{A}) , along a feasible path $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(x_0, \tilde{h}^{(0)})$, with associated history $\tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}),$

$$\forall t, F(\tilde{h}^{(t)}, x_t, x_{t+1}) \le a_1(x_0)a^t + a_2$$

Then for all $T \in \mathbb{N}$,

$$\sum_{t=0}^{T} \beta^{t} F(\tilde{h}^{(t)}, x_{t}, x_{t+1}) \leq \sum_{t=0}^{T} \beta^{t} (a_{1}(x_{0})a^{t} + a_{2})$$
$$= a_{1}(x_{0}) \sum_{t=0}^{T} (a\beta)^{t} + a_{2} \sum_{t=0}^{T} \beta^{t}$$

since $0 < a\beta < 1$ and $0 < \beta < 1$ the conclusion follows.

6.2 **Proof of Proposition 2.2**

Let us show that the objective is upper semi-continuous and the feasible sequence set is a compact set of the product topology. By (F), one can check by induction that for any given $k_0 \in \mathbb{R}^+$ and $\tilde{h}^{(0)} \in l^{\infty}_+$, for all feasible sequence $\tilde{x} = (x_t)_t \in \Pi(\tilde{h}^{(0)}, x_0)$, for all $t, x_{t+1} \in \Gamma(\tilde{h}^{(t)}, x_t)$, we have

$$\forall t, x_t \le a^t ||x_0|| + \frac{1 - a^t}{1 - a}a' = (||x_0|| - \frac{a'}{1 - a})a^t + \frac{a'}{1 - a}$$

The feasible set $\Pi(\tilde{h}^{(0)}, x_0)$ is included in a compact set for the product topology. Moreover, it is closed. So the feasible set $\Pi(\tilde{h}^{(0)}, x_0)$ is also compact.

We next show that the objective function \mathcal{U} is upper semi-continuous.

Let us consider a sequence $\tilde{x}^n = \{(x_t^n)_{t=1}^{+\infty}\}_n \subset \Pi(\tilde{h}^{(0)}, x_0)$ that converges to $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(\tilde{h}^{(0)}, x_0)$. Note that when *n* converges to $+\infty$, by (m1), the sequence of associated histories $\forall t \ge 1, \tilde{h^n}^{(t)} = (m(x_{t-1}^n, x_t)^n, m(x_{t-2}^n, x_{t-1}^n), \dots, m(x_1^n, x_2^n), m(x_0^n, x_1^n), \tilde{h}^{(0)})$ converges to the associated history $\forall t \ge 1, \tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)})$.

Let us show that $\lim_{n \to +\infty} \mathcal{U}(\tilde{x}^n) \leq \mathcal{U}(\tilde{x})$. The notation \lim means \limsup .

For any $t \ge 0$, by (A2), for any $\tilde{x} = (x_t)_{t=1}^{+\infty} \in \Pi(\tilde{h}^{(0)}, x_0)$ and with history $\tilde{h}^{(t)}$ associated to \tilde{x}

$$F^+(\tilde{h}^{(t)}, x_t, x_{t+1}) \le a_1(x_0)a^t + a_2$$

and by $0 < a\beta < 1$, for any $\varepsilon > 0$, there exists T_{ε} such that for any $(x_t)_{t=1}^{+\infty} \in \Pi(\tilde{h}^{(0)}, x_0)$, and for any $T \geq T_{\varepsilon}$,

$$\sum_{t=T}^{+\infty} \beta^t F^+(\tilde{h}^{(t)}, x_t, x_{t+1}) \le \varepsilon$$

So for any $\varepsilon > 0$, there exists T_{ε} such that for any $n \in \mathbb{N}$ and for any $T \ge T_{\varepsilon}$,

$$\sum_{t=T}^{+\infty} \beta^t F^+(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) \le \varepsilon$$

and for any $n \in \mathbb{N}$ and for any $T \geq T_{\varepsilon}$,

$$\begin{split} \sum_{t=0}^{+\infty} \beta^t F(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) &= \sum_{t=0}^T \beta^t F(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) + \sum_{t=T}^{+\infty} \beta^t F(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) \\ &\leq \sum_{t=0}^T \beta^t F(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) + \sum_{t=T}^{+\infty} \beta^t F^+(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) \\ &\leq \sum_{t=0}^T \beta^t F(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) + \varepsilon \end{split}$$

By taking $n \to +\infty$ (and using the continuity of F in the right-hand side of the above inequality),

$$\lim_{n \to +\infty} \sum_{t=0}^{+\infty} \beta^t F(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) \le \sum_{t=0}^T \beta^t F(\widetilde{h}^{(t)}, x_t, x_{t+1}) + \varepsilon$$

Since this is true for any $T \ge T_{\varepsilon}$, by taking $T \to +\infty$,

$$\lim_{n \to +\infty} \sum_{t=0}^{+\infty} \beta^t F(\widetilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) \le \sum_{t=0}^{+\infty} \beta^t F(\widetilde{h}^{(t)}, x_t, x_{t+1}) + \varepsilon$$

Since this is true for any $\varepsilon > 0$, by taking $\varepsilon \to 0$,

$$\lim_{n \to +\infty} \sum_{t=0}^{+\infty} \beta^t F(\tilde{h^n}^{(t)}, x_t^n, x_{t+1}^n) \le \sum_{t=0}^{+\infty} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1})$$

So \mathcal{U} is upper semi-continuous on $\Pi(\tilde{h}^{(0)}, x_0)$.

By Weierstrass Theorem (Aubin[4], Theorem 5.3.1), since \mathcal{U} is upper semi-continuous and $\Pi(\tilde{h}^{(0)}, x_0)$ is a compact set for the product topology, there exists an optimal solution.

The assumptions that F is jointly strictly concave ensures the uniqueness of the solution.

6.3 Proof of Proposition 2.3

A direct proof using (A2) can be done. Indeed, let us consider a sequence $(\widetilde{h^n}^{(0)}, x_0^n)_n \subset l_+^{\infty} \times X$ that converges to $(\widetilde{h}^{(0)}, x_0) \in l_+^{\infty} \times X$, use the fact that x_0^n converges to x_0 and let us consider a subsequence $(\widetilde{h^{n_i}}^{(0)}, x_0^{n_i})_i$ such that

$$\lim_{n \to +\infty} V(\widetilde{h^{n}}^{(0)}, x_0^n) = \lim_{i \to +\infty} V(\widetilde{h^{n_i}}^{(0)}, x_0^{n_i})$$

Let $\varepsilon > 0$. By (A2), there exist i_0 and T_0 such that for any $i \ge i_0$ and for any $T \ge T_0$, and for optimal path $(\tilde{x}^{n_i})_i \in \Pi(\widetilde{h^{n_i}}^{(0)}, x_0^{n_i})$ and its associated history $\widetilde{h^{n_i}}^{(t)}$,

$$V(\widetilde{h^{n_i}}^{(0)}, x_0^{n_i}) = \sum_{t=0}^{+\infty} \beta^t F(\widetilde{h^{n_i}}^{(t)}, x_t^{n_i}, x_{t+1}^{n_i}) \le \sum_{t=0}^T \beta^t F(\widetilde{h^{n_i}}^{(t)}, x_t^{n_i}, x_{t+1}^{n_i}) + \varepsilon$$

Fix $T \ge T_0$. The subsequence $(\tilde{x}^{n_i})_i$ that belongs⁶ to $\Pi(\widetilde{h^{n_i}}^{(0)}, x_0^{n_i})$ can be assumed to converge to some \tilde{x} in $\Pi(\tilde{h}^{(0)}, x_0)$. By the definition of the associated history and the continuity of m, this implies that $(\widetilde{h^{n_i}}^{(t)})_i$ converges to $\tilde{h}^{(t)}$ the history associated to \tilde{x} . Let $i \to +\infty$, by the continuity of F,

$$\lim_{n \to +\infty} V(\widetilde{h^{n}}^{(0)}, x_0^n) \le \sum_{t=0}^T \beta^t F(\widetilde{h}^{(t)}, x_t, x_{t+1}) + \varepsilon$$

Let $T \to +\infty$,

$$\lim_{n \to +\infty} V(\tilde{h^n}^{(0)}, x_0^n) \le \mathcal{U}(\tilde{x}) + \varepsilon \le V(\tilde{h}^{(0)}, x_0)$$

by the arbitrariness of ε .

6.4 Proof of Proposition 2.4

One can check (see Le Van and Morhaim[34]) that (F2) and (A) imply that (H) $\forall x_0 \in X, \exists \mathcal{V}(x_0)$ a compact neighborhood of x_0 in $X, \forall \varepsilon > 0, \exists T_0$ such that $\forall T \ge T_0, \forall x'_0 \in \mathcal{V}(x_0), \forall \tilde{x}' \in \Pi(\tilde{h}^{(0)}, x'_0)$, one has with $\tilde{h}'^{(t)} = (m(x'_{t-1}, x'_t), m(x'_{t-2}, x'_{t-1}), \dots, m(x'_1, x'_2), m(x'_0, x'_1), \tilde{h}^{(0)})$,

$$\sum_{t=T}^{+\infty} \beta^t F^+(\tilde{h}'^{(t)}, x'_t, x'_{t+1}) \le \varepsilon$$

where $F^+(h, r, r') = \max\{0, F(h, r, r')\}.$

(i) By (H), $\exists T_0, \forall T > T_0, \forall x'_0 \in \mathcal{V}(x_0), \varepsilon > 0, \forall \tilde{x}' \in \Pi(\tilde{h}^{(0)}, x_0),$

$$\sum_{t=T}^{+\infty} \beta^t F(\tilde{h}'^{(t)}, x'_t, x'_{t+1}) \le \sum_{t=T}^{+\infty} \beta^t F^+(\tilde{h}'^{(t)}, x'_t, x'_{t+1}) \le \varepsilon$$

⁶ by the compactness of $\Pi(\widetilde{h^{n_i}}^{(0)}, x_0^{n_i})$

Let $\tilde{x}' \in \Pi(\tilde{h}^{(0)}, x'_0), T \ge T_0$. For any $\tilde{x}'' = (x''_{T+1}, ...) \in \Pi(\tilde{h}'^{(T)}, x'_T)$, one has $(x'_1, ..., x'_T, x''_{T+1}, ...) \in \Pi(\tilde{h}^{(0)}, x'_0)$, and $\beta^T F(\tilde{h}'^{(T)}, x'_T, x''_{T+1}) + \beta^{T+1} F(\tilde{h}'^{(T+1)}, x''_{T+1}, x''_{T+2}) + \le \varepsilon$

so $\beta^T V(\tilde{h}^{(T)}, x'_T) \leq \varepsilon$ which implies (i).

 $(ii) \ \forall \tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0),$

$$-\infty < \mathcal{U}(\tilde{x}) \le \sum_{t=0}^{T} \beta^{t} F(\tilde{h}^{(t)}, x_{t}, x_{t+1}) + \beta^{T+1} V(\tilde{h}^{(T+1)}, x_{T+1})$$

and

$$0 = \lim_{T \to +\infty} [\mathcal{U}(\tilde{x}) - \sum_{t=0}^{T} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1})] \le \lim_{T \to +\infty} \beta^{T+1} V(\tilde{h}^{(T+1)}, x_{T+1})$$

The notation <u>lim</u> means liminf. From (i) then $\lim_{T\to+\infty} \beta^{T+1} V(\tilde{h}^{(T+1)}, x_{T+1}) = 0$

6.5 **Proof of Proposition 2.6**

The proof that V is a fixed-point of the Bellman operator is standard (see Stokey Lucas and Prescott[49]). Uniqueness of the fixed point is shown by contradiction. Indeed, suppose there exists W another fixed-point of \mathcal{B} in $\mathcal{F}_b(l^{\infty}_+ \times X, \mathbb{R})$. Let us first check that $W \leq V$. Let $(\tilde{h}^{(0)}, x_0)$ be given. There exists $x_1 \in \Gamma(\tilde{h}^{(0)}, x_0)$ such that $W(\tilde{h}^{(0)}, x_0) = F(\tilde{h}^{(0)}, x_0, k_1) + \beta W(\tilde{h}^{(1)}, x_1)$ and by induction, there exists a sequence $(x_t)_{t\geq 1}$ with associated history sequence $(\tilde{h}^{(t)})_t$ such that for any T,

$$W(\tilde{h}^{(0)}, x_0) = \sum_{t=0}^{T-1} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1}) + \beta^T W(\tilde{h}^{(T)}, x_T)$$

Since W belongs to $\mathcal{F}_b(l^{\infty}_+ \times X, \mathbb{R})$, one has taking the limit when $T \to +\infty$, and then by V being the sup of the sum

$$W(\tilde{h}^{(0)}, x_0) \le \sum_{t=0}^{+\infty} \beta^t F(\tilde{h}^{(t)}, x_t, x_{t+1}) \le V(x_0, \tilde{h}^{(0)})$$

Let us now show that $V \leq W$. Let $x_0 \in X$, $\tilde{h}^{(0)} \in l_+^\infty$. For any $\tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0)$ such that $\mathcal{U}(\tilde{x}) > -\infty$, one has, with $\tilde{h}^{(1)} = (m(x_0, x_1), \tilde{h}^{(0)})$,

$$\begin{split} W(\tilde{h}^{(0)}, x_0) &= BW(\tilde{h}^{(0)}, x_0) \\ &\geq F(\tilde{h}^{(0)}, x_0, x_1) + \beta W((m(x_0, x_1), \tilde{h}^{(0)})), x_1) \\ &= F(\tilde{h}^{(0)}, x_0, x_1) + \beta W(\tilde{h}^{(1)}, x_1) \end{split}$$

and so by induction, with $\tilde{h}^{(t)} = (m(x_{t-1}, x_t), m(x_{t-2}, x_{t-1}), \dots, m(x_1, x_2), m(x_0, x_1), \tilde{h}^{(0)}),$

$$W(k_{0}, \tilde{h}^{(0)}) \geq \sum_{t=0}^{T} F(\tilde{h}^{(t)}, x_{t}, x_{t+1}) + \beta^{T+1} W(\tilde{h}^{(T+1)}, x_{T+1})$$

$$\geq \lim_{T \to +\infty} \sum_{t=0}^{T} F(\tilde{h}^{(t)}, x_{t}, x_{t+1}) + \lim_{T \to +\infty} \beta^{T+1} W(\tilde{h}^{(T+1)}, x_{T+1})$$

$$= \mathcal{U}(\tilde{x})$$

which implies that $W(\tilde{h}^{(0)}, x_0) \ge V(\tilde{h}^{(0)}, x_0)$ (since for any $\tilde{x} \in \Pi(\tilde{h}^{(0)}, x_0)$, one has $W(\tilde{h}^{(0)}, x_0) \ge \mathcal{U}(\tilde{x})$ and $V(\tilde{h}^{(0)}, x_0)$ is the sup of $\mathcal{U}(\tilde{x})$ for \tilde{x} in $\Pi(\tilde{h}^{(0)}, x_0)$).

Finally, this shows that dynamic programming tools can be used to deal with general historydependent optimal growth models.

References

- Emilio Abad-Segura, Ana Batlles de la Fuente, Mariana-Daniela González-Zamar, and Luis Jesús Belmonte-Ureña. Effects of circular economy policies on the environment and sustainable growth: Worldwide research. Sustainability, 12(14):5792, 2020.
- [2] Daron Acemoglu and Matthew O Jackson. History, expectations, and leadership in the evolution of social norms. *The Review of Economic Studies*, 82(2):423–456, 2015.
- [3] Mikael Skou Andersen. An introductory note on the environmental economics of the circular economy. *Sustainability science*, 2(1):133–140, 2007.
- [4] Jean-Pierre Aubin. Initiation à l'analyse appliquée. Masson Paris, Milan, Barcelone, 1994.
- [5] Mauro Bambi, Daria Ghilli, Fausto Gozzi, and Marta Leocata. Habits and demand changes after covid-19. *Journal of Mathematical Economics*, 110:102933, 2024.
- [6] Mauro Bambi and Fausto Gozzi. Internal habits formation and optimality. Journal of Mathematical Economics, 91:165–172, 2020.
- [7] Manel Baucells and Rakesh K Sarin. Predicting utility under satiation and habit formation. Management Science, 56(2):286–301, 2010.
- [8] Lawrence E Blume and Daniel L Rubinfeld. The dynamics of the legal process. The Journal of Legal Studies, 11(2):405–419, 1982.
- [9] Kenneth E Boulding. The economics of the coming spaceship earth. In *Environmental quality* in a growing economy, pages 3–14. RFF Press, 1966.
- [10] Christopher D Carroll, Jody Overland, and David N Weil. Comparison utility in a growth model. Journal of economic growth, 2(4):339–367, 1997.
- [11] Christopher D Carroll, Jody Overland, and David N Weil. Saving and growth with habit formation. *American Economic Review*, 90(3):341–355, 2000.
- [12] Thiago AC de Melo, Marcelo A de Oliveira, Sara RG de Sousa, Raimundo K Vieira, and Thayane S Amaral. Circular economy public policies: A systematic literature review. *Procedia Computer Science*, 204:652–662, 2022.
- [13] Kieran P Donaghy. 17. getting to a circular growth economy by harnessing circular and cumulative causation. Handbook on Entropy, Complexity and Spatial Dynamics: A Rebirth of Theory?, page 287, 2021.
- [14] Kieran P Donaghy. A circular economy model of economic growth with circular and cumulative causation and trade. *Networks and Spatial Economics*, 22(3):461–488, 2022.

- [15] Karen E Dynan. Habit formation in consumer preferences: Evidence from panel data. American Economic Review, 90(3):391–406, 2000.
- [16] Daniel A Farber. The rule of law and the law of precedents. Minn. L. Rev., 90:1173, 2005.
- [17] Oscar Fitch-Roy, David Benson, and David Monciardini. All around the world: Assessing optimality in comparative circular economy policy packages. *Journal of cleaner production*, 286:125493, 2021.
- [18] Jeffrey C Fuhrer. Habit formation in consumption and its implications for monetary-policy models. American Economic Review, 90(3):367–390, 2000.
- [19] Donald AR George, Brian Chi-ang Lin, and Yunmin Chen. A circular economy model of economic growth. Environmental modelling & software, 73:60–63, 2015.
- [20] Patrizia Ghisellini, Catia Cialani, and Sergio Ulgiati. A review on circular economy: the expected transition to a balanced interplay of environmental and economic systems. *Journal of Cleaner* production, 114:11–32, 2016.
- [21] John A Goldsmith and Bernard Laks. *Battle in the mind fields*. University of Chicago Press, 2019.
- [22] Tomas Havranek, Marek Rusnak, and Anna Sokolova. Habit formation in consumption: A meta-analysis. European Economic Review, 95:142–167, 2017.
- [23] Ying He, James S Dyer, and John C Butler. On the axiomatization of the satiation and habit formation utility models. *Operations Research*, 61(6):1399–1410, 2013.
- [24] Ying He, James S Dyer, and John C Butler. A decision-making model with utility from anticipation and disappointment. Journal of Multi-Criteria Decision Analysis, 26(1-2):35–50, 2019.
- [25] Geoffrey M Heal and Harl E Ryder. Optimal growth with intertemporally dependent preferences. The Review of Economic Studies, 40(1):1–31, 1973.
- [26] Masako Ikefuji. Habit formation in an endogenous growth model with pollution abatement activities. *Journal of Economics*, 94(3):241–259, 2008.
- [27] Yuliya Kalmykova, Madumita Sadagopan, and Leonardo Rosado. Circular economy-from review of theories and practices to development of implementation tools. *Resources, conservation and recycling*, 135:190–201, 2018.
- [28] Myrto Kasioumi. The environmental kuznets curve: Recycling and the role of habit formation. *Review of Economic Analysis*, 13(3):367–387, 2021.
- [29] Myrto Kasioumi. Economics of Recycling. PhD thesis, University of Guelph, 2022.
- [30] Myrto Kasioumi and Thanasis Stengos. A circular model of economic growth and waste recycling. *Circular Economy and Sustainability*, pages 1–26, 2022.
- [31] Myrto Kasioumi and Thanasis Stengos. The environmental kuznets curve under recycling and habit formation. *Working Paper*, 2022.

- [32] Julian Kirchherr, Nan-Hua Nadja Yang, Frederik Schulze-Spüntrup, Maarten J Heerink, and Kris Hartley. Conceptualizing the circular economy (revisited): An analysis of 221 definitions. *Resources, Conservation and Recycling*, 194:107001, 2023.
- [33] Cuong Le Van and Lisa Morhaim. Optimal growth models with bounded or unbounded returns: a unifying approach. *Journal of economic theory*, 105(1):158–187, 2002.
- [34] Cuong Le Van and Lisa Morhaim. Optimal growth models with bounded or unbounded returns: a unifying approach. *Journal of Economic Theory*, 105(1):158–187, 2002.
- [35] Sebastian Lewis. Precedent and the rule of law. Oxford journal of legal studies, 41(4):873–898, 2021.
- [36] Åsa Löfgren. Habit formation in the environmental quality: dynamic optimal environmental taxation. rapport nr.: Working Papers in Economics, (92), 2003.
- [37] Ellen Foundation MacArthur. How the circular economy tackles climate change. *Ellen MacArthur Found*, 1:1–71, 2019.
- [38] Asaf Mazar, Geoffrey Tomaino, Ziv Carmon, and Wendy Wood. Habits to save our habitat: Using the psychology of habits to promote sustainability. *Behavioral Science & Policy*, 7(2):75– 89, 2021.
- [39] Vincent Moreau, Marlyne Sahakian, Pascal Van Griethuysen, and François Vuille. Coming full circle: why social and institutional dimensions matter for the circular economy. *Journal of Industrial Ecology*, 21(3):497–506, 2017.
- [40] Lisa Morhaim and Ayşegül Yıldız Ulus. On history-dependent optimization models: a unified framework to analyze models with habits, satiation and optimal growth. *Journal of Mathematical Economics*, page 102807, 2023.
- [41] David W Pearce and R Kerry Turner. *Economics of natural resources and the environment*. Johns Hopkins University Press, 1989.
- [42] Enrichetta Ravina. Habit formation and keeping up with the joneses: Evidence from micro data. 2019.
- [43] Sofia Ritzén and Gunilla Olundh Sandström. Barriers to the circular economy-integration of perspectives and domains. Procedia Cirp, 64:7–12, 2017.
- [44] Kareen Rozen. Foundations of intrinsic habit formation. *Econometrica*, 78(4):1341–1373, 2010.
- [45] Aldo Rustichini and Paolo Siconolfi. Dynamic theory of preferences: Habit formation and taste for variety. *Mimeo*, 2005.
- [46] Aldo Rustichini and Paolo Siconolfi. Dynamic theory of preferences: Habit formation and taste for variety. *Journal of Mathematical Economics*, 55:55–68, 2014.
- [47] Fatma Safi and Lobna Ben Hassen. Subtractive versus multiplicative habits in environmental economics. *Economic Alternatives*, (1):72–90, 2021.
- [48] Sjak Smulders, Michael Toman, and Cees Withagen. Growth theory and green growth. Oxford review of economic policy, 30(3):423–446, 2014.

- [49] Nancy L Stokey, RE Lucas, and E Prescott. Recursive methods in dynamic economics. Cambridge, MA: Harvard University, 1989.
- [50] Guy Tennenholtz, Nadav Merlis, Lior Shani, Martin Mladenov, and Craig Boutilier. Reinforcement learning with history dependent dynamic contexts. In *International Conference on Machine Learning*, pages 34011–34053. PMLR, 2023.
- [51] Ayşegül Yıldız Ulus. Pontryagin's principle for a class of discrete time infinite horizon optimal growth problems. In *Mathematical Modelling and Optimization of Engineering Problems*, pages 51–69. Springer, 2020.
- [52] Cuong Van and Rose-Anne Dana. Dynamic programming in economics, volume 5. Springer Science & Business Media, 2003.
- [53] Frederick Van Der Ploeg and Cees Withagen. Pollution control and the ramsey problem. *Environmental and Resource Economics*, 1(2):215–236, 1991.
- [54] Alexander Vostroknutov. Social norms in experimental economics: Towards a unified theory of normative decision making. Analyse & Kritik, 42(1):3–40, 2020.